



Configurations and Applications of Multi-Agent Hybrid Drone/Unmanned Ground Vehicle for Underground Environments: A Review

Chris Dinelli¹, John Racette¹, Mario Escarcega¹, Simon Lotero², Jeffrey Gordon², James Montoya¹, Chase Dunaway¹, Vasileios Androulakis², Hassan Khaniani³, Sihua Shao⁴, Pedram Roghanchi² and Mostafa Hassanalian^{1,*}

- ¹ Department of Mechanical Engineering, New Mexico Tech, Socorro, NM 87801, USA
- ² Department of Mineral Engineering, New Mexico Tech, Socorro, NM 87801, USA
- ³ Petroleum Recovery Research Center, New Mexico Tech, Socorro, NM 87801, USA
- ⁴ Department of Electrical Engineering, New Mexico Tech, Socorro, NM 87801, USA
- * Correspondence: mostafa.hassanalian@nmt.edu

Abstract: Subterranean openings, including mines, present a unique and challenging environment for robots and autonomous exploration systems. Autonomous robots that are created today will be deployed in harsh and unexplored landscapes that humanity is increasingly encountering in its scientific and technological endeavors. Terrestrial and extraterrestrial environments pose significant challenges for both humans and robots: they are inhospitable and inaccessible to humans due to a lack of space or oxygen, poor or no illumination, unpredictable terrain, a GPS-denied environment, and a lack of satellite imagery or mapping information of any type. Underground mines provide a good physical simulation for these types of environments, and thus, can be useful for testing and developing highly sought-after autonomous navigation frameworks for autonomous agents. This review presents a collective study of robotic systems, both of individual and hybrid types, intended for deployment in such environments. The prevalent configurations, practices for their construction and the hardware equipment of existing multi-agent hybrid robotic systems will be discussed. It aims to provide a supplementary tool for defining the state of the art of coupled Unmanned Ground Vehicle (UGV)-Unmanned Aerial Vehicle (UAV) systems implemented for underground exploration and navigation purposes, as well as to provide some suggestions for multi-agent robotic system solutions, and ultimately, to support the development of a semi-autonomous hybrid UGV-UAV system to assist with mine emergency responses.

Keywords: subterranean autonomous exploration; drones; bio-inspired robots; coupled UGV–UAV systems; multi-agent hybrid robotic systems

1. Introduction

The ever-increasing demands of society for material resources are leading scientific and industrial societies to explore and exploit harsh environments. Terrestrial and extraterrestrial environments pose significant challenges for both humans and robots: they are inhospitable and inaccessible to humans due to a lack of space or oxygen, poor or no illumination, unpredictable terrain, a GPS-denied nature (Global Positioning System), and a lack of satellite imagery or mapping information of any type [1–4]. Underground mines and openings are a good physical simulation of these types of environments, and thus, can be useful for testing and developing highly sought-after autonomous navigation frameworks for robots [5,6]. A variety of dynamic and static obstacles, such as machinery and heavy equipment, infrastructure, mine workers, falling material, and suspending dust, prevalent in mines during normal mining operations are additional complexity factors [7]. Furthermore, during mining emergency situations, catastrophes can create even more complex



Citation: Dinelli, C.; Racette, J.; Escarcega, M.; Lotero, S.; Gordon, J.; Montoya, J.; Dunaway, C.; Androulakis, V.; Khaniani, H.; Shao, S.; et al. Configurations and Applications of Multi-Agent Hybrid Drone/Unmanned Ground Vehicle for Underground Environments: A Review. *Drones* **2023**, *7*, 136. https:// doi.org/10.3390/drones7020136

Academic Editor: Diego González-Aguilera

Received: 5 January 2023 Revised: 10 February 2023 Accepted: 12 February 2023 Published: 14 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). obstacles and compounding dangers in these environments, which, to name a few, are: flash fires, massive shaft or roof collapses, water, a shortage of oxygen, wind, and water. Thus, mining emergencies can worsen the hazardous conditions for humans inside the mine, as well as block the escape paths and completely trap them.

In such emergencies, specialized first responders are deployed to rescue the trapped miners, so long as the mine is safe enough for the team to enter and effectively execute their mission. These missions are not always successful, and even upon the rescue of the trapped mine workers, a fully successful outcome is not guaranteed. Sending these search and rescue teams inside a mine during these emergencies requires the first responders to risk their lives and well-being. Requiring human responders to voluntarily enter an inhospitable, dynamically changing, hazardous, and potentially deadly environment with no guarantee of a safe return is an inefficient solution.

Naturally, integrating the recent scientific and technological advances in such missions can play a critical role in increasing the chances for successful outcomes and saving many lives [5–7]. Equipping these search and rescue teams with advanced tools and frameworks by harnessing the rapid technological advances in robotics, computer hardware, and artificial intelligence is the most sensible next step in the creation of future mines. The use of robots for improving safety and efficiency in underground spaces is not a new concept, although a new tool is at the forefront of this technology. Autonomous, unmanned robotic agents are independent robotic elements that are capable of user-independent operation and problem solving, hence rendering human supervision obsolete. Removing the human operator from the loop means that these robots can perform missions in environments that are either inaccessible or unsafe for human agents. Furthermore, while a single autonomous agent can accomplish many tasks independently, there are some constraints that necessitate a combination of agents working in tandem. In some cases, the combination of multiple robotic agents can dramatically increase the system's overall efficiency by eliminating the risk of a single-point failure, which stems from the complete reliance on a single independent agent.

Conceptually, these autonomous and semi-autonomous multi-agent systems have as much variation and as wide a range of applications as those in the modern mechatronics and robotics fields do. The advance of collaborative swarms of UGVs and drones that are able to operate, navigate, communicate, and even install and assemble "each other" as a symbiotic team will play a critical role in the industry, science, and society in the future [8]. Highly durable and technologically robust multi-agent hybrid robotic systems (MAHRSs) could provide solutions not only to modern problems and mysteries, but also to ones that humanity has yet to encounter. Multi-agent systems that require little to no human input are a future solution to many of the infrastructure issues that modern humans face today. Specifically, when they are applied to subterranean and other GPS-denied environments, these systems can provide an inherently modular and flexible solution to the dynamic and unpredictable obstacles that are present. The application of these systems to modern problems will have long-standing effects on the safety, flexibility, and efficiency of accessing some of the harshest environments on our planet and far beyond its limits.

The MAHRSs discussed in this review are deployed exclusively in GPS-denied environments with applications ranging from mine safety to archaeology and excavation to structural health monitoring. As the design of these robotic systems becomes more robust in terms of intrinsic safety, mission efficiency, and operational time optimization, future terrestrial and extraterrestrial exploration missions will grant us access to the landscapes and terrains that humans cannot currently reach. To understand the future applications of these systems, one must first examine how science and industry are shaping the landscape of robotics in underground spaces today.

This review focuses primarily on unmanned ground and aerial vehicles, either as stand-alone agents or as multi-agent systems (hybrid) in underground spaces, and especially, fully GPS-denied ones. The principal motivation for this study stems from the desire to improve the safety and health of mine rescue teams deployed during mine emergencies. The impact of smart, multi-agent systems on mitigating the exposure of human rescuers to the occupational hazards prevalent in a mine during an emergency can potentially be enormous. The same smart systems can be deployed in similarly harsh environments, terrestrial or extraterrestrial ones, which are inaccessible or inhospitable to humans. Therefore, the scope of this study is to provide a comprehensive review of the design elements and the operation constraints of existing robotic systems. To understand the applications of autonomous robots in these environments, one must evaluate the applications of their mechanical predecessors, the present state-of-the-art ones, and the envisioned future applications. Hence, this study examines UGVs and drones that have already been designed and deployed in underground spaces. This means that some aspect their configuration (size, shape, locomotion type, etc.) has been optimized for these environments. This review also discusses bio-inspired robots that have not, but could potentially, be optimized for auxiliary individual or swarm operations. The ultimate scope of this review is to serve as a collection of literature reviews, Mineral Engineering R&D, and Intelligent Robotic Systems R&D, which aim to improve and expedite future R&D on intelligent underground systems, further understand the impact of the technological advancements upon human safety and work efficiency in harsh terrestrial and extraterrestrial environments, and finally, to inspire designers and innovators of such systems.

The rest of the paper is organized as follows: Section 2 discusses extensively the fundamental classification of unmanned systems, with the emphasis given to Unmanned Ground Vehicles (UGVs) and Unmanned Aerial Vehicles (UAVs), as well as hybrid UAV–UGV (single-agent systems). Section 3 focuses on cooperative UGV–UAV systems (multi-agent systems). In Section 4, a discussion of mines and underground hazards and challenges stemming from the nature of these environments is presented, while Section 5 provides detailed underground applications of hybrid multi-agent systems. Finally, Section 6 discusses the main takeaway from this review and presents some conclusions.

2. Classification of Unmanned Systems

As considered in this review, MAHRSs can be described as the coupling of two or more robotic systems to complete missions as a team. This concept can be compared to biological symbiosis, defined as "any type of a close and long-term biological interaction between two different biological organisms, be it mutualistic, commensalistic, or parasitic" [9]. While the roles of each robot within the system vary in complexity and importance, the overarching objective of using a multi-part hybrid robotic system is to utilize the specializations of many robotic designs, instead of creating one robot that can accomplish every task as an individual. Therefore, each robot is designed to accomplish a mission-specific objective that is part of a bigger objective. Hence, the term MAHRS encompasses not only the design and function of the individual robotic agents, but also the interfacing, connectivity, and collaboration rules between the robotic agents and the simultaneous (passive/active) operation of these multiple specialized robotic agents in order to design a holistically efficient system. It is vital to utilize this design concept to understand the state-of-the-art UGVs and drones as they can be found in commercial marketplaces, scientific publications, industrial settings, and military applications. To begin with, one must first become familiar with the fundamental classification of the different robots, the nomenclature that exists to define these systems, and the individual functionality of each classification. In the following subsections, single robot systems are reviewed, and specifically UGVs, UAVs (or drones), and single hybrid UGV–UAV robots (see Figure 1).



Figure 1. Classification of unmanned systems for underground applications.

2.1. Unmanned Ground Vehicles (UGV)

Multiple configurations of unmanned ground vehicles have been developed and implemented for various purposes. In this subsection, a review of UGVs is presented with emphasis on wheeled, tracked, as well as bio-inspired robots. In the latest category, a brief review of legged, snake-like, earthworm-inspired, climbing, rolling, jumping, and hybrid motion robots is discussed.

2.1.1. Wheeled Vehicles

Wheeled UGVs are one of the most common types of terrestrial robots deployed for environment exploration, whether it is a well-structured environment or an unknown, harsh environment [10]. Commonly, a UGV is designed to execute a specialized mission based on the output of its exploration and mapping module (see Figure 2). Wheel-based designs come in a wide range of configurations, from the widely used four-wheeled designs to highly complex multi-wheeled rovers designed to traverse highly uneven landscapes [11]. The size, design, and type of the wheels can vary based on the working space and the specialized missions assigned to the UGV. Four-wheeled designs, which are mechanically simpler, enable efficient locomotion on relatively even surfaces. This is the reason that the four-wheeled configuration prevails in most passenger vehicles that are in operation today. By applying torque to the center of each wheel, the robots can move forwards and backward, while by manipulating the steering angle of the wheels or their adjoining shafts, the robots can turn, spin, flip, or perform similar operations. Autonomous sensing and perception capabilities in modern robotics mean that these systems, such as most of the systems discussed in this review, can be autonomously operated by defining a small number of inputs.

Wheeled robots have been used in many scientific fields of research, industrial purposes, and military applications [12]. International manufacturers and distributors make various commercial platforms that are available in the marketplace. These platforms can be equipped with a wide variety of sensors, mechanical arms and handlers, and artificial intelligence to aid processes in nearly every industry and application. However, implementing a terrestrial robot to solve a given problem requires researchers or inventors to select between constructing a customized robot or utilizing an existing commercial robot. The choice must be made upon careful consideration of the intended implementation and the specialized missions that will be assigned to the robot. The literature is abundant on



such specialized robots, with applications ranging from agricultural monitoring and pruning (see Figure 2a) to extraterrestrial exploration (see Figure 2b).

Figure 2. Examples of wheeled UGVs with specialized functionalities: (**a**) four-wheeled UGV drives through seedlings, picking weeds between rows [13]; (**b**) six-wheeled UCV traverses Mars' surface.

The movement of a wheeled robot, and thus its capabilities in each working space, is determined by the number and configuration of the wheels and the wheel axles. For example, a two-wheeled robot with differential drive (see Figure 3a) has three degrees of freedom (DOF), i.e., the position (x, y) and the heading (φ) and two actuators, i.e., the motors on the two wheels [13]. These motors enable the robot to move forwards and backward and rotate in place, but the robot cannot move laterally without first rotating. In comparison, a four-wheeled robot with Ackermann steering has the same three DOF, but two different actuators, i.e., a motor and a steering wheel (see Figure 3b). The motor enables the robot to move forwards and backward, while the steering wheel can only orient the former movement. As a result, the four-wheeled robot can neither rotate in place nor move laterally [13].



Figure 3. Accessible and non-accessible DOF for differential drive and Ackermann steering robots [13].

When one is constructing a customized wheeled UGV, the design will be focused upon a number of determinative characteristics that include, but are not limited to: the number of wheels, the power source of torque (electric batteries, combustion engines, environmental energy harvesting techniques, etc.), the wheel position on the robot's frame, the orientation of the wheels during normal operation, removability, retractability, the wheels' multi-part design (rim and tire combination), material selection, a factor of safety for failure, size, weight, the center of gravity, and the wheel shape (sphere, cylinder, abstract shape, etc.).

Wheel placement and orientation on the platform change how the actuation of the wheels affects the overall motion. Four-wheeled designs with differential drives, such as the one seen in Figure 4, use a simple design to move forward and backward. However, dynamic equations can model the movement of a robot's movement and the effect of various positions and orientations of the wheels. These models can give useful insights into the various ways to elicit unconventional locomotion, such as the case of the three-wheeled robot depicted in Figure 4.



Figure 4. A three-wheeled platform that can actuate each motor about a central axis [13].

An important benefit of utilizing or fabricating wheeled platforms stems from the popularity of these systems. The infinite information and resources from previous implementations and in-depth research and experimentation can be significantly insightful [14]. As one of the oldest and most efficient forms of mechanical locomotion, four-wheeled systems are used today for many applications, and thus, have been readily integrated with odometry sensors and techniques and kinematic analysis that can help to improve various aspects of navigation algorithms (see Figure 5a–c). Despite the inherent operational simplicity of the four-wheeled systems, they are unable to efficiently traverse uneven or slippery terrain, which reduces the necessary wheel-to-ground friction.



Figure 5. Views of (**a**) Husky UGV from Clearpath Robotics equipped with two robotic arms (Clearpath Robotics [15]), (**b**) commercially available wheel-based UGV equipped with LiDAR (SLAM/LIDAR.drone.online), and (**c**) Husky UGV from Clearpath Robotics equipped with custom fabricated soil manipulator crane arm [16].

2.1.2. Tracked Vehicles

Tracked UGVs provide a robust alternative to wheeled UGVs [17]. By spreading the weight of the vehicle across a larger surface area, more friction-based adherence to the ground is created. This allows these types of vehicles to operate in uneven and soft-soiled terrain. Although they are slower in speed, the benefits of increased grip and stability are coupled with a stable and simple drive mechanism that allows the tracked vehicles to rotate around themselves (skid-steer locomotion). [18] In most platforms, a track-based

UGV consists of two tracked wheel systems on either side, each with multi-directional control, as seen in Figure 6.



Figure 6. Track-based UGVs are pre-manufactured and sold by commercial distributors both online and in stores. (Walmart, Online Marketplace, Toys and Electronics, 2022).

Track-based suspension is often installed in tracked vehicles to mitigate the impact of ground-to-track interactions. The installation of spring-mass systems or dampeners allows the tracked vehicles to navigate steep or sudden obstacles without damaging the overall drive mechanisms or the sensor modalities on board (see Figures 7 and 8) [19].



Figure 7. The spring-loaded suspension between tank track drives and supports [19].



Figure 8. Researchers equip a Husky UGV from Clearpath Robotics with tank tracks that fit over the original UGV wheels [16].

Morphing tracks are tank tracks on a UGV that can change their shape or orientation (see Figure 9). The timing of the change in orientation is often up to the designer of the robot, but this capability to change and morph shapes instantaneously gives these drone agents more tools and degrees of freedom to overcome complex obstacles [20].



Figure 9. View of a morphing-track based robot [20].

2.1.3. Bio-Inspired Robots

Bio-inspired robots commonly borrow locomotive mechanisms from the natural world to achieve efficient navigation techniques in environments that are similar to the ones that encouraged the natural evolution of living organisms. Most of the bio-inspired robots are small in size, and hence, they do not have the overload capabilities to support complex autonomous functionalities. However, these robots can be used for auxiliary subtasks of an MAHRS's overall mission. For example, the micro-robots could be used for deploying or reconstructing wireless communication network grids, creating distinctive landmarks or beacons to assist the navigation of the collaborating robotic agents, creating networks for data stations (such as meteorological or hydrological stations), or even assisting in the self-evacuation of trapped humans in subterranean environments. The small size of these robots allows them to carry a single sensor or signaling beacon, while at the same time, it facilitates the storage and deployment of these robots from the larger collaborating robotic agents. Moreover, as nanotechnology, solid state technology, and MEMS advance such robots are becoming increasingly more powerful. The great flexibility of such dynamically deployed networks could be a critical factor for the successful execution of an MAHRS's mission. The bio-inspired robots examined here include legged robots, snakelike, earthworm-inspired, climbing, rolling, jumping, as well as hybrid motion robots.

2.1.3.1. Legged Robots

Bio-inspired robots borrow many mechanisms from nature, but one very common means of locomotion that is used is leg-based propulsion. Leg-based designs can be used for walking, running, or even jumping [21–24]. Leg-based designs can contain any number of legs based on the animals that are simulating, e.g., bipedal ones, quadrupedal ones, hexapods ones, etc. (see Figure 10). Each design achieves a slightly different locomotive goal depending on how many legs are present and how they are arranged.

A spring-loaded scissor jack leg uses multiple joints to accomplish the greater distribution of load across the leg; when they are placed in line and operated by a tank-driven motor, bio-inspired legs can be used even in tank track configurations (see Figure 11) [19].

There are also many approaches to the design of an individual leg. One common realization of a robotic leg includes double-jointed assemblies (one at the hip and one at the knee) [25]. This joint distribution is similar to the leg designs of humans, dogs, cats, etc., ignoring the joint at the ankle. The double-jointed design can be realized with different actuators, both linear and rotational ones, as shown in Figure 12. The benefit of using a linear actuator as the knee joint, known as the pogo stick method, is the simplicity of the system. In this case, there is no geometric work due to the presence of a single angle of the action. The double-rotary joint design, however, is more closely related to the legs of animals. While the design is complicated by the second angle that is introduced, there are more stances, gaits, and types of motion possible with this added dimension of flexibility than there are in the pogo stick design. In Figure 12, views of a robotic leg with one linear



actuator at the knee and a rotary actuator at the hip and a leg with two rotary actuators, one at the hip and one at the knee, are shown [25].

Figure 10. View of leg-based robots.



Figure 11. Views of (a) scissor jack leg assembly and (b) scissor jack tank suspension assembly [19].



Figure 12. Views of (**a**) a robotic leg with one linear actuator at the knee and a rotary actuator at the hip and (**b**) a leg with two rotary actuators: one at the hip and one at the knee [25].

2.1.3.2. Snake-Like Robots

Among snakes, four primary locomotion modes, also called gaits, have been documented: lateral (serpentine) undulation, concertina, sidewinding, and rectilinear motion [26] (see Figure 13). Concertina locomotion involves a longitudinal wave throughout the snake's body, which winds up in a sort of ribbon shape, and then extends forward, and the snake repeats the process. Serpentine locomotion is the gait that is most associated with snake movement. During this type of locomotion, the body generates a transverse wave from head to tail and uses this periodic oscillation to propel itself forward. Sidewinding locomotion is the fastest way for a snake to travel, as it involves the sideways stretching and contraction of the body, where only a couple of points on the snake's body contact the ground at a time. This type of locomotion is often likened to a sort of 'running' motion in snakes that are capable of it because of their speed and a small amount of relative ground contact. The last type of snake locomotion, rectilinear or caterpillar locomotion, is a linear mode of transport. This locomotion is best used in cramped environments where sideways undulation is not possible due to the restricted space. Longitudinal stretching and compression make this slow locomotion method possible in the same way that a caterpillar moves, with subtle undulating movements [26].



Figure 13. Views of four primary modes of snake locomotion: concertina, serpentine, sidewinding, and rectilinear [26].

Different gaits are desired for different engineering goals when one is designing robots with snake-like locomotion properties. Sidewinder locomotion is likely desired for applications where the robot is expected to speedily transverse a loose, sandy environment or an environment where the ground surface is dangerously hot or otherwise corrosive due to short contact duration. Rectilinear locomotion is necessary for locomotion in cramped environments such as inside slender tunnels or pipes [27]. The types of applications for this type of locomotion include search and rescue missions in tight spaces or pipe inspection for non-destructive testing purposes. Serpentine motion has the advantage of being relatively easy to model using simple actuator units. The locomotion takes advantage of traveling transverse waves, and thus, it can be achieved by creating mechanical oscillations across the body of the robot. This type of bio-inspired serpentine motion is used by the robot shown in Figure 14 [27].



Figure 14. Snake-inspired robot with passive wheels controlled by non-linear oscillators which mimic serpentine locomotion [27].



In 2013, Raytheon Sarcos developed a new generation tandem track snake-like robot (see Figure 15), which is suitable for operations in underground mines [28].

Figure 15. Snake-inspired robot developed by Raytheon Sarco for operations in underground mines [28].

2.1.3.3. Earthworm-Inspired Robots

Earthworms utilize longitudinal undulation in order to propel themselves forward across surfaces and underground. Their narrow bodies are well adapted to this type of motion, and building robots whose movements are inspired by earthworms has promising applications such as search and rescues, industrial inspections, and medical endoscopy [29].

In order to replicate this method of locomotion, small, linear actuators must be used to replicate longitudinal undulation. As evident in Figure 16, shape memory alloy (SMA) wires are coiled into spring shapes and linked together, then wrapped in a nylon mold. When an electric current is applied to the brass linkages in this SMA skeleton, the spring is heated to induce the Joule effect. This causes the SMA spring to contract, and when the current is shut off, the spring is able to expand gradually, and this process is repeated to create periodical, longitudinal undulating motion [29].



Figure 16. View of (**a**) shape-memory alloy skeleton of robotic earthworm design and (**b**) wrapped in a Nylon mold [29].

This SMA skeleton, which is wrapped in the nylon casing, creates the robotic earthworm capable of undulatory travel with four independently controllable modules. Undulatory patterns with a typical frequency of 0.5 Hz is used to create a longitudinal wave that travels across the length of the robot and allows it to propagate through a medium or across a surface. The use of tiny micro-legs on the base of the design allows it to generate enough traction to climb up inclined surfaces with small undulations (see Figure 17) [29].



Figure 17. View of robotic earthworm design using shape memory alloy locomotion technology (a) with and (b) without micro-legs [29].

2.1.3.4. Climbing Robots

Many natural climbing locomotion instances can be found and studied when one is intending to design a robotic climber. Many animals, such as insects, lizards, and amphibians, can easily climb sheer surfaces. Much of their climbing success can be attributed to the natural adhesive effects of their appendages, their lightweight, and the efficiency of their motion. Geckos, for instance, utilize directionally preferential microfibers on their foot pads to generate a frictional force on vertical surfaces, which counteracts its own gravitational force [30]. This is made possible by preloading each foot placement and dragging downwards to generate adhesion. In order to detach each foot pad, gradual peeling is performed to nullify the adhesion in a similar fashion to that of hook and loop fashioners such as VELCRO[®]. This process is depicted in Figure 18 [30].



Figure 18. Views of (**a**) ideal attaching and detaching phase diagram of gecko foot used while climbing and (**b**) gecko-inspired climbing robot with flexible foot and hip joints and Silly Putty[®] adhesive foot pads [30].

When smaller, faster robots are required, bio-inspiration may be drawn from the insect or arachnid classes of the animal kingdom. In order to maximize the effectiveness of the climbing robot, considerations need to be made as to what makes their natural counterparts such effective climbers. A high contact-to-weight ratio is desired for adhesion, which is achieved by a low mass and a high leg count. This allows insects such as cockroaches to be very efficient and very fast climbers. Micro-actuators and shape memory alloys can be used to replicate insect-based climbing in robotic applications, such as the one shown in Figure 19 [31].

A four-legged climbing robot "Limbed Excursion Mechanical Utility Robot" called LEMUR was designed and prototyped by NASA's Jet Propulsion Laboratory and several other institutions. LEMUR uses grippers that are armed with hundreds of tiny, sharp hooks that enable it to climb rock walls and navigate underground spaces [32]. In Figure 20, a view of this climbing robot is shown.



Figure 19. Views of (**a**) climbing insect, (**b**) insect-inspired robot, (**c**) numerical insect model, and (**d**) numerical robot model. The numerical models were used to compare normal reaction forces and investigate leg configuration, trajectory, and compliances [31].



Figure 20. View of a four-legged climbing robot [33].

2.1.3.5. Rolling Robots

Some robots are designed to be able to move via rolling, which can be especially effective and energy saving when they are traversing certain terrains where gravity aids the direction of travel. Some of them are designed to be completely globular, relying on rolling as the sole method of maneuvering. Rolling robots that use rolling as their sole locomotion method can be classified into three control categories: control by the center of mass, control by variable gyrostatic momentum, and control by deformation [34]. Other hybrid designs are designed to be able to walk and roll, which makes them more similar to the animals they draw inspiration from, such as isopods, armadillos, and rolling spiders (see Figure 21). These robots often involve a process that contracts them into a spherical shape, which is a process called conglobation.



Figure 21. View of (**a**) rolling isopod-based robot demonstrating the conglobation process [34] and (**b**) golden-wheel-inspired rolling robot [24].

2.1.3.6. Jumping Robots

Some bio-inspired robots attempt to emulate the jumping motion of many animal species, such as frogs and kangaroos. This method of locomotion may be beneficial to the robot when it is navigating terrain with obstacles that are impossible to traverse with a traditional rolling wheel locomotion method. One of the difficulties of the jumping method of locomotion is designing a robot that remains perfectly balanced throughout each consecutive jump, and extensive testing is needed to find the robot's center of gravity. Many jumping robots, such as the kangaroo bot shown in Figure 22, utilize movable counterbalances to simulate the tail. This keeps the robot from over-rotating or under-rotating during each jump and sets it up to proceed with the subsequent jump [35].



Figure 22. View of a jumping-based bio-inspired kangaroo robot; (**a**) transmission system of kangaroo robot, (**b**) CAD drawing of kangaroo robot, and (**c**) photo of kangaroo robot [35].

2.1.3.7. Hybrid Motion Bio-Inspired Robots

While many animals have achieved the mastery of certain types of motion, there are also many animals that are capable of multiple modes of transportation. Whether it is climbing and walking, walking and swimming, climbing and flying, or many other combinations, there are natural counterparts to observe for any desired type of locomotion one might have for robotic design. One of the most desirable qualities of a locomotive robot is the ability to traverse both land and water. These amphibious robots can be modeled with inspiration from many amphibious animals, such as otters, crocodilians, salamanders, marine iguanas, and many more. Figure 23 shows a prototype for one such amphibious robot design. Instead of utilizing swimming and walking locomotion modes such as salamanders or crocodilians do, this prototype design uses actuated fins and a wheel that doubles as a propeller when it is underwater [36].



Figure 23. Mechanism design of AmphiRobot-II; prototype of the amphibious robot in the field test [36].

15 of 54

2.2. Drones

Recently, the mining industry has shown an increased interest in using drones for routine operations in surface and underground mines [4–6]. Shahmoradi et al. provided a comprehensive review of the current state of drone technology and its applications in the mining industry. Their study presented the configurations, specifications, and applications of commercially available drones for mining applications [4]. In the following subsections, different drone configurations are discussed including single- and multi-rotors, thrust-vectoring aerial vehicles, moving-mass controlled vehicles, coandă effect drones, flapping, winged drones, airships/blimps, and cyclocopters.

2.2.1. Single Rotors/Helicopters

Single-rotor drones are useful in the applications of surveying and construction. Due to the greater battery efficiency of powering a single propeller, mono-rotors can lift heavier payloads than multi-rotor drones of the same weight can (see Figure 24). This also creates faster drones, although at the expense of flight stability and directional control fidelity. The biggest challenge to designing a drone with a single rotor is controlling the horizontal flight direction. Using a single rotor enables very easy and efficient vertical flight, but translating itself sideways in a specific direction is difficult to achieve when only a single rotor is used [1,37].



Figure 24. Single-rotor helicopter-style drone used for surveillance [37].

2.2.2. Ducted Fan/Thrust Vectoring

The challenge of achieving directional horizontal maneuverability can be addressed by utilizing thrust vectoring. Thrust vectoring is a process by which an aerial vehicle can manipulate the direction of thrust from its motor. A mono-rotor drone can accomplish this by including actuated fins below the propeller, which alter and divert the airflow in the desired direction. The mono-rotor drone shown in Figure 25 utilizes this technique and a ducted fan to enable high-speed, controllable, multi-directional flight using a single propeller. The ducted fan enables a more efficient, higher velocity thrust from the propeller by housing it in a cylindrical shroud [1,3,38].



Figure 25. Design of single rotor drone, complete with ducted propeller, four actuated fins, and a structural/landing support [38].

2.2.3. Moving-Mass Controlled (MMC) Drones

Moving-mass controlled (MMC) drones utilize a unique method of directional control that involves changing the drone's center of gravity (see Figure 26). Unlike other single-rotor drones, which often use swiveling tilt rotors to control the horizontal movement of the drone, MMC drones often have a fixed rotational axis for the propeller and instead use linear shifts in weight for directional control. Linear actuators, often moving in two spatial dimensions, provide the required shift in the center of mass to achieve the desired directional flight change [1,39].



Figure 26. Single-rotor drone controlled by two-dimensional moving masses which direct the drone's motion by controlling the center of gravity [39].

2.2.4. Coandă Effect Drones

The Coandă effect, named after the Romanian inventor Henri Coandă, is the tendency of a fluid jet, such as air, to stay attached to a convex surface [1]. This effect can be seen when one is observing the fluid flow around a curved airfoil, causing the flow direction to be altered by the convex surface geometry [40,41]. An illustration of this effect can be seen in Figure 27.



Figure 27. Illustration of the Coandă effect to produce resultant lift as air passes through the center of an airfoil ring and increases velocity as it travels around [41].

This effect is used for certain designs that alter the airflow direction and speed for cooling, such as the Dyson Air MultiplierTM fan, or lift, such as the Aesir Coandă drone shown in Figure 28 [41].



Figure 28. Aesir drone utilizing a single rotor and the Coandă effect for flight [41].

2.2.5. Multi-Rotor Drones

Besides single-rotor drones or drones that move without rotors, several designs of drones adopt two or more rotors that enable better directional control. In the following subsections, a brief overview of bi-rotors, tricopters, quadcopters, and pentacopters is presented.

2.2.5.1. Bi-Rotors

As opposed to the more conventional quadcopter design, it is possible to design a bi-rotor drone with tilting rotors to enable directional control. The primary reason that quadcopters are used more often than bi-rotor drones is the appeal of utilizing fixed rotors while still being able to maneuver the drone forwards and backwards, left and right, as well as making it rotate. The bi-rotor does not have this flexibility with fixed rotors, and balance and symmetry must be sacrificed when one is opting the two-rotor configuration. It is worth noting that this configuration still provides more stability with fewer workarounds than the mono-rotor configuration does [1].

One of the main advantages of using a bi-rotor drone design over a quadcopter one is battery efficiency. The tilting mechanisms on each rotor enable the drone to be controlled without much loss of control or stability, but powering only two rotors facilitates significantly longer flight times, nearly doubling the maximum flight duration from about a half hour (as typical in quadcopter flights) to fifty minutes in the case of the V-Coptr Falcon by Zero Zero Robotics, as shown in Figure 29 [42].



Figure 29. V-Coptr Falcon by Zero Zero Robotics with two tilting rotors [42].

2.2.5.2. Tricopters

Tricopters, such as the one shown in Figure 30, are drones with three rotors as opposed to four or more. These drones are vastly less popular than quadcopters are for a few key reasons. While reducing the number of rotors from four to three may reduce the weight of the drone, the loss of thrust imposes limitations. The tricopter has a longer battery life than quadcopters do due to the need to power one less rotor, but not by a large enough margin to make up for the many downsides to three-rotor control. The main reason that tricopters have a hard time competing with quadcopters is the difficulty of control arising from the frame's asymmetry. While quadcopters have four-way symmetry and intuitive solutions to controlling pitch, yaw, and roll, the tricopter does not. Due to the asymmetry of the tricopter, flight control is much more complex than it is with the quadcopter. As a result, numerous studies have been conducted to devise mechanical solutions to controlling altitude, pitch, yaw, and roll. Altitude control and pitch control are relatively straightforward with the tricopter, but unusual combinations of rotor power and servo motors are required to control the drone's roll and yaw. As an example of these complexities, some controlling techniques are illustrated in Figure 31 [1,43].



Figure 30. Tri-rotor drone in flight [43].



Figure 31. Tri-rotor drone control strategies for (a) altitude; (b) roll; (c) pitch; (d) yaw [44].

2.2.5.3. Quadcopter

Quadcopters with four individually controlled rotors are the most widely used design for drones. The popularity of this design comes from the ease of control granted by the two axes of symmetry, allowing for pitch, yaw, and roll to be controlled with ease. While the battery efficiency is less appealing compared to the drones with less than four rotors, most commercial quadcopters are often able to sustain flights in the range of thirty minutes. Figure 32 shows a custom-made quadcopter based on a commercial racing frame kit. Common applications for commercial quadcopters include security and surveillance, topographical mapping, emergency responses, infrastructure inspections, and even drone racing [45].



Figure 32. Semi-autonomous precision-docking quadcopter constructed from a racing drone frame with custom sensors and flight controllers [45].

2.2.5.4. Pentacopters and Others

While quadcopters are considered to be the standard drone design for most applications, drones with a higher rotor count exhibit advantages for specific applications despite the disadvantages of an increased weight, size, and cost of the drone, as well as the decrease in battery life and flight time. Drones with a large number of rotors come with the benefit of a higher weight-lifting potential. Although the weight of multi-rotor drones is larger, the added rotors enable it to have a higher overload capacity. As a result, hexacopter and octocopter drones are often used in applications that require them to lift larger payloads, including delivery drone applications such as the hexacopter shown in Figure 33 [1,46].



Figure 33. Hexacopter drone used for package delivery [46].

While four-, six-, and eight-rotor drones have many commercial applications, fiverotor pentacopters are very rarely used. This is due to the greatly increased difficulty of piloting a drone with an odd number of symmetries, and the development of piloting algorithms for such symmetries lag compared to those of other drones. Some pentacopters utilize quadrilateral symmetry, with the fifth rotor placed in the design center (see Figure 34a)



or used as a frontal propeller (Figure 34b), which allows them to be piloted in a very similar way to traditional quadcopters [1].

Figure 34. Various pentacopter designs: (**a**) central upwards rotor on experimental drone design [47], (**b**) front-facing propeller on Foxtech Screamer Racing Pentacopter [48], (**c**) experimental 'dragonfly' design with four front rotors and a tail for pitch control [49], and (**d**) a purely hypothetical equidistant rotor design [50].

Although pentacopters are often deemed to be impractical for flight due to the difficulty of controlling them, six-rotor and eight-rotor drones are commercially available for purposes such as surveying and payload delivery (see Figure 35). One of the benefits of sixand eight-rotor drones, besides their higher lift potential, is rotor redundancy. This means that in the event of a rotor failure during flight, the drone can remain airborne, as opposed to a quadcopter, which becomes inoperable in the event of a single rotor failure [1].



Figure 35. Multi-rotor drones include commercial versions of (**a**) a hexacopter by Sky Rider [51] and (**b**) an octocopter by DJI [52].

2.2.6. Flapping Wings

Flapping wing drones are inspired by birds or insects and imitate them with the help of mechanical wings to gain altitude and fly in the desired manner (see Figure 36). Flapping wing drones have the potential to offer better maneuverability advantages over those of other drones of the same size [53–63]. Small insect- and hummingbird-inspired drones could potentially be capable of Vertical Takeoff and Landing (VTOL). A disadvantage of flapping wing drones is the complexity of their actuation mechanism.



Figure 36. A flapping wing drone that took bio-inspiration from dragonflies [59].

2.2.7. Airships and Blimps

Unmanned airships rely on lighter-than-air buoyancy for lift, as opposed to rotors or flapping wings. These drones are required to be much larger than traditional quadcopters or flapping wing drones are due to the requirement of large volumes of lifting gas (see Figure 37). A major advantage of airships is that they require a very small amount of battery power to stay gas aloft, and hence, they endure for longer. On the other hand, airships have the major drawback of a low payload capacity despite their size. The large size of airships also creates a lot of drag, which hinders their performance in air currents and prevents them from achieving high speeds [1,64].



Figure 37. An autonomous airship that uses thrusters and tail fins for control [65].

2.2.8. Cyclocopters

Cyclocopters generate lift through an arrangement of airfoils that rotate continuously to control lift and thrust (see Figure 38). Configuring the pitch and rotation of the airfoils enable the control of the cyclocopter's movement [66]. Cyclocopters can come in several configurations, including twin and quad configurations, and they are also capable of VTOL [1,3]. However, two major drawbacks of cyclocopters are their mechanical complexity and the complexity of their control.





2.3. Hybrid Drone–UGV Systems

Hybrid single-agent systems where a UGV can convert to or simultaneously function as a drone, and vice versa, are a popular alternative to expand the capabilities of a single robot. Common examples include tank quadcopters, walking/jumping drones, rolling–flying drones, climbing–flying drones, flying UGVs, and transformable drones.

2.3.1. Tank Quadcopters

One type of unmanned hybrid aerial-terrestrial vehicle is a tank quadcopter, i.e., a quadcopter fitted with powered tank treads, allowing for both all-terrain ground travel and aerial travel. This option allows the unmanned vehicle to navigate and execute missions in a large range of environments. The treads would allow the drone to easily traverse difficult terrain and could be designed to be relatively lightweight compared to a drone with multiple wheels. Treads that wrap all the way around the drone, such as in Figure 39, would also allow the drone to continue ground travel upside down if the vehicle happens to be turned over.



Figure 39. Views of tank quadcopter complete with powered treads for terrestrial travel and four rotors for aerial travel, (**a**) [67], (**b**) [68], (**c**) [69], and (**d**) [70].

2.3.2. Walking/Jumping Drones

Walking and jumping drones are both potential concepts for hybrid aerial–groundbased robots (see Figure 40). They would both allow the drone an alternative means of locomoting on the ground; this allows the drones to traverse terrain without flying, which can save the battery power. A drawback to walking/jumping drones is that the drone has to use additional battery power to lift the added weight of the walking/jumping mechanism [71].





2.3.3. Rolling–Flying Drones

Another interesting concept for a hybrid aerial–ground-based robot is a rolling–flying drone such as the one in Figure 41. It is similar in concept to a flying UGV; however, this concept focuses more on the drone aspect than it does on the UGV aspect. A rolling drone is advantageous because it has the ability to roll on terrain, whenever possible, which allows it to use less battery power while it is travelling. A drawback of rolling–flying drones is that flying requires more power because it has to lift the additional weight of the rolling mechanism.



Figure 41. (a) Parrot MiniDrone Rolling Spider is an example of a quadcopter drone that utilizes both flying and rolling mechanics of motion, fitted with wheels that allow it to roll across the ground and up walls [72] and (b) Tilt Ranger drone with V-SCAN3D scanner (Clickmox Solutions) [73,74].

2.3.4. Climbing-Flying Drones

Climbing–flying drones are designed with the ability to climb obstacles or walls if they landed on or near them. The climbing functionality allows the drones to climb over difficult terrain or obstacles instead of flying over the obstacles, and hence, it reduces energy consumption. VertiGo, shown in Figure 42, uses a bi-rotor configuration to assist the drone with climbing walls [75].



Figure 42. VertiGo, a drone developed by Disney's research team, is an example of a drone that is capable of scaling walls by using four gripping wheels and a bi-rotor drone, which allows flight and maintains traction on walls [75].

2.3.5. Flying UGVs

Flying UGVs are another possible hybrid aerial–ground system. Unlike rolling–flying drones or tank quadcopters, the UGV is the primary focus of this type of transformable autonomous agent (see Figure 43). The flight capability of the UGV allows the agent to fly over terrain that is impassable to the UGV's locomotion mechanism. However, a major drawback of a flying UGV is the higher power consumption due to the rotors having to lift the entire weight of the UGV. This can be mitigated by keeping the UGV's flights short in duration.



Figure 43. Advanced Tactics Panther Drone is one of the first unmanned ground vehicles (UGVs) capable of flight in addition to terrestrial travel. The UGV uses four powered wheels, as well as four rotors that act as a quadcopter with pitch and roll control [76].

2.3.6. Transformable Drones

Transformable robots are designed to be able to alter their configuration or shape. Many of them consist of smaller modular components that can be assembled in various manners, as shown in Figure 44. Others have hinges or sections that can move dynamically, allowing the drone to change shape. The ability to transform can be beneficial since it allows the drone to adapt its configuration according to the unique needs of the environment in which the drone operates, allowing it to adapt to dynamic environments.



Figure 44. View of (**a**) Airblock transformable drone is an example of a modular transformable drone that consists of a central core and six modular rotor blocks, which can be rearranged into different flight and ground travel modes, as seen above [77], (**b**) DRAGON, which is somehow an acronym for the tongue-twisting description "Dual-Rotor Embedded Multilink Robot with the Ability of Multi-Degree-of-Freedom Aerial Transformation," [78], (**c**) this drone can transform into a tiny car to slide under small gaps [79], and (**d**) morphing design for quadrotors [80].

3. Cooperative UGV-UAV Systems

There have been recent efforts to increase the degree of automation and frequency of data collection for GPS-denied environment applications using UGV–UAV systems. The combination of drones and UGV, where a drone can be deployed by a UGV, and vice versa, can be applied for search and rescue missions in harsh environments. However, the current practice of the missions in these environments is traditionally performed, which is manual, costly, and time consuming. Developing vision-based mobile robotic systems that are aware of their surroundings and capable of autonomous/semi-autonomous navigation are becoming essential to many search and rescue applications, specifically for underground environments.

Applying a single unmanned vehicle in an underground environment mission incurs a series of performance issues. One major problem is inefficient navigation and agility in indoor and cluttered spaces with many obstacles and barriers, where some places are inaccessible by a UGV. To provide a solution to this problem, a team of mobile robots which integrate drones and UGVs can be applied. In these systems, an unmanned ground vehicle will semi-autonomously/autonomously navigate underground spaces while carrying a drone. The drone will act as an external eye for the UGV, observing the scenes in underground environments that are inaccessible to the UGV.

3.1. Drones Deployed by UGVs

One possibility for drone and UGV integration is to have drones that are deployed by UGVs. This option is advantageous because it allows the UGV to carry drones to the mission; this allows the drone to have a longer flight duration in the mission area because the drone does not have to use its battery while it is in transit to the mission area. The UGV can also act as a charging station for deployed drones; UGVs equipped with housing for drones protect the drone from the elements. Drones deployed by UGVs can also aid with the UGV's operations. The drone can be used to assist with mapping obstacles for the UGVs navigation system. The drone can also be used to directly assist the UGV with navigating obstacles. One system that was created used a drone to tether a UGV's cable to difficult terrain; the UGV was then able to use a winch system to pull itself over difficult terrain [81].

Planck AeroSystems and Stealth Technologies have developed a capability for autonomous UGV that could deploy drones autonomously. The UGV has been developed for perimeter security applications in sectors such as defense, transport, government and utilities, and energy (see Figure 45a) [82]. A Russian combat robot also has been designed for patrolling without human assistance, navigating its way across a 100 km route and working with a swarm of drones (see Figure 45b) [83]. The robotic dog/grasshopper from Boston Dynamics has been modified to carry and deploy a drone (see Figure 45c) [84]. In 2014, a team of researchers from the University of Waterloo developed a control system that enabled a drone to land safely on a UGV (see Figure 45d) [85]. Cocchioni et al. worked on the interaction between a UGV and a drone to extend the endurance of the aerial system thanks to a novel landing/recharging platform (see Figure 45e) [86].



Figure 45. Views of drones deployed by UGVs; (**a**) UGV-drone system developed by Planck AeroSystems and Stealth Technologies [82], (**b**) Russian combat robot working with a swarm of drones [83], (**c**) robotic dog/grasshopper from Boston Dynamics carries and deploys a drone [84], (**d**) University of Waterloo's control system that enabled a drone to land safely on a UGV [85], (**e**) UGV-drone interaction developed by Cocchioni et al. [86], (**f**) heterogenous UGV–drone system developed by Saska et al. [87], (**g**) visual-aided autonomous takeoff, tracking, and landing of a drone on a moving UGV by Palafox et al. [88], (**h**) a cooperative exploration using a UGV and a drone proposed by Hood et al. [89], (**i**) a drone and an UGV was examined for detailed inspections of power lines by Cantieri et al. [90], and (**j**) integrated a blimp drone and a Husky UGV tested by Asadi et al. [91].

Saska et al. presented a heterogenous UGV–drone system that could cooperatively solve periodical surveillance tasks in indoor environments (see Figure 45f) [87]. Palafox et al., in 2019, performed the visual-aided autonomous takeoff, tracking, and landing of a drone on a moving UGV (see Figure 45g) [88]. Hood et al. proposed a cooperative ex-

ploration using a UGV and a drone. In this research, the UGV could navigate through the free space, and the drone provided enhanced situational awareness via its higher vantage point. The goal of this application was perform search and rescue in a damaged building (see Figure 45h) [89]. In the paper by Cantieri et al., an architecture involving a drone and an unmanned ground vehicle (UGV) was examined for detailed inspections of power lines (see Figure 45i) [90]. Asadi et al. designed and tested a drone–UGV team that integrated a blimp drone and a Husky UGV (see Figure 45j) for construction environments [91].

3.1.1. Tethered Drones Deployed by UGVs

Another application for integrating drones and UGVs is a tethered drone system. This system involves having a drone that is directly tethered to the UGV with cables and wires. A major advantage of the system is that the drone can directly pull power from the UGV's main power supply; this relieves the UGV from carrying the drone-charging system. The tethered drone assists with the UGV's mapping. On the other hand, the major disadvantage of this system is the tether that limits the range of the drone. Another disadvantage of tethering the drone to the UGV is the potential for the tether to become tangled, while the tether also could affect the maneuverability of the UGV. In Figure 46a–c, views of tethered drones deployed by military UGVs are shown [92].



Figure 46. Views of tethered drones deployed by military UGVs [92].

In 2014, Papachristos and Tzes worked on autonomous cooperation between a tethered drone and a UGV to improve navigation in partially mapped environments (see Figure 47a) [93]. Miki et al., in 2019, proposed a cooperative system for a drone and a UGV, which uses the drone not only as a flying sensor, but also as a tether attachment device that assists the UGV in climbing a cliff (see Figure 47b) [81]. Nguyen and Garone proposed a system composed of a drone and a UGV, which cooperated to manipulate and move an object to the desired pose [94]. Xiao et al., in 2020, developed an autonomous tethered drone in conjunction with a UGV for operations in unstructured or confined environments (see Figure 47c) [95]. Viegas et al. in 2021, described the design, prototyping, and testing of a novel, lightweight, tethered drone with a mixed multi-rotor design and water jet propulsion for forest fire fighting (see Figure 47d) [96].

In 2022, Martinez-Rozas et al. addressed the problem of trajectory planning in a marsupial robotic system consisting of a drone connected to a UGV through a non-taut tether that has a controllable length. This work aimed to determine a synchronized collision-free trajectory for the three marsupial system agents, including the drone, UGV, and tether [97]. Papachristos et al. presented a complete framework for developing a tethered drone deployed by a ground robot for search and rescue operations (see Figure 47e) [98]. Mullens et al. developed a UGV-mounted automated refueling system for VTOL drones (see Figure 47f) [99]. In 2022, Borgese et al. worked on a multi-robot system composed of a UGV and a tether-connected drone to carry out missions in outdoor environments (see Figure 47g) [100]. Quaglia et al. developed a solar-powered UGV that carries a drone for precision agriculture (see Figure 47h) [101]. Gu et al. proposed a tethered system including roaming tethered aerial robots for radioactive material sampling and stationary tethered



aerial robots for environmental monitoring to meet extremely long endurance missions in nuclear power plants (see Figure 47i) [102].

Figure 47. Views of tethered drones deployed by UGVs, (**a**) autonomous cooperation between a tethered drone and a UGV developed by Papachristos and Tzes [93], (**b**) cooperative system for a drone and a UGV proposed by Miki et al. [81]., (**c**) an autonomous tethered drone in conjunction with a UGV developed by Xiao et al. [95], (**d**) a tethered drone with a mixed multi-rotor design and water jet propulsion for forest fire fighting designed by Viegas et al. [96], (**e**) a tethered drone deployed by a ground robot for search and rescue operations presented by Papachristos et al. [98], (**f**) UGV-mounted automated refueling system for VTOL drones developed by Mullens et al. [99], (**g**) a UGV and a tether-connected drone designed by Borgese et al. [100], (**h**) a solar-powered UGV that carries a drone for precision agriculture developed by Quaglia et al. [101], and (**i**) a tethered system including roaming tethered aerial robots for radioactive material sampling and stationary tethered aerial robots for environmental monitoring proposed by Gu et al. [102].

3.1.2. Housing Platforms for Drones on UGVs

Drone docking stations allow the drones to take off and land and provide a recharging opportunity for them. These stations are generally designed for battery-powered multirotor and VTOL drones. For the drones that are deployed by UGVs, the housing mechanism is a key feature of any UGV that houses drones in extreme environments such as underground spaces. An opening and closing hatch is an advantageous type of housing because it is a relatively simple mechanism that protects the housed drone from the elements. The opening and closing hatch mechanisms are also advantageous because they can prevent debris from accumulating on the landing pad while the drone is deployed. Another advantage of the opening and closing hatch mechanism is the potential for opening the hatch to clear debris that may have accumulated on top of the hatch. Another type of housing mechanism is an opening and closing IRIS; IRIS designs are advantageous because the mechanism does not risk hitting an external object while it is opening. A disadvantage of the IRIS mechanism is the possibility of debris getting stuck on the IRIS while it is closed; this could lead to debris falling on the drone when the IRIS opens for deployment. Figures 48–50 show some of the commercially available housing systems for drones.



Figure 48. Views of (**a**) DBX-G7 drone station by H3 Dynamics [103], (**b**) AERIEPORT by Volatus Aerospace [103], (**c**) stabilized drone landing platform by STABLE [103], (**d**) ENCATA Engineering Catalyst Autonomous drone station [104], (**e**) Singaporean Dronebox, a solar-powered landing station for miniature aircraft [105], (**f**) Valqari's drone delivery station [106], (**g**) Strix Drones, drone agnostic docking stations for advanced operations [107], (**h**) Matternet docking station [108], (**i**) Mobotix drone station [109], (**j**) DJI Dock [110], (**k**) Percepto's Sparrow I drone base station (Courtesy) [111], (**l**) SkyX drone-charging stations [112], (**m**) Hextronics global advanced drone dock [113], and (**n**) Hextronics Atlas 300 docking station [113].



Figure 49. Views of (**a**) IDIPLOYER nexus dock [113], (**b**) Heisha D80 dock [113], (**c**) Heisha D135 drone dock [113], (**d**) CounterDome drone station [113], (**e**) Airscort FlytMini [113], (**f**) Aerobox docking station [113], (**g**) Hive Droneport docking station [113], (**h**) Matternet drone station [114], (**i**) 3D Delivery drone station rigged by 3d_molier International [115], (**j**) Airmada automated drone ground stations [116], (**k**) STRIX 1600 drone station [117], (**l**) STRIX 2100 drone station [117], and (**m**) Vtrus ABI Zero drone and base station [118].

As shown in Figures 48–50, these structures house drones in order to protect and charge them. Similar concepts can be designed for the UGV to house and hold the drone before deployment in underground spaces and protect the electrical components from dust. This structure should be fabricated as a box with an opening/closing hatch and shock absorbers to protect and hold the drone during the motion of the UGV.



Figure 50. Views of (**a**) Autel automatic charging station for EVO II drones [119], (**b**) Avy Dock [120], (**c**) DHL drone dock [121], (**d**) SKEYETECH drone dock [122], (**e**) Airobotics battery-swapping platform [123], (**f**) American Robotics drone station [124], (**g**) Antwork UGV drone dock [125], (**h**) Skycharge Skyport DP5 [126], (**i**) Foxit Response [127], (**j**) Airscort Flytmini [128], (**k**) Airscort [128], (**l**) ARGOSDYNE drone station [129], (**m**) John Deere drone station [130], and (**n**) Singapore Technologies Aerospace [131].

3.1.3. Charging Mechanisms for Drones on UGVs

Drone-charging systems can be attached to UGVs to allow them to charge the integrated drones that they house. Drone-charging stations make use of either wired or wireless methods of charging the drone's battery. Wired methods traditionally use a wired contact that can make contact with the drone's battery to charge it or replace the battery. Although wired contact can provide very efficient charging, it requires precise landing, and it poses a danger from the exposed electrical contacts. An autonomous maintenance station should be able to provide a means for guiding drones to land, along with a replacement of the depleted battery with a fully charged battery. The goal of having a battery replacement mechanism in drone docks/housings is to decrease the downtime between the charging of drone batteries in applications, such as smart cities, border patrol, military, medical delivery, and postal systems. The specific drones that are typically used are multi-rotors, with typically three or more rotors to produce lift.

Currently, the greatest challenge to overcome with drones is the limited battery capacity and the down time between each charge. There are various methods to keep drones functioning, but some of them are more feasible than others. The proposed solutions include drones outfitted with solar panels, using fossil fuels as the energy source, a combination of solar- and laser-powered drones, and an autonomous battery replacement system. Once, these solutions were evaluated, it is clear the standard limitations of drones become more significant in the design. Other limitations include the total mass, size, maximum altitude, number of propellers, and payload delivery. When one is considering these limitations, it becomes clear that the mass of the drone will most likely make other approaches ineffective, and this leaves only the option of autonomous battery replacement.

One of the first studies was conducted by Fujii et al. [132], where they found success in replacing drone batteries for indoor and outdoor drones. In their design, they illustrated a need for a landing platform, position correction, and battery swapping. Underneath the landing platform was a hidden battery that was preloaded and charged, and once the drone was centered, the battery was then loaded into the drone [132]. In 2015, Nutting and Mercik [133] from Worcester Polytechnic Institute also worked on a similar problem. In this work, they focused more on the locking mechanism of the battery and how it would make contact with the terminals of the drone. They cleverly designed a battery housing with rails on the sides so that the battery could slide into the drone. Their drone and battery case included neodymium magnets as the terminals for powering the drone [133]. As the first two studies were focused on landing stations with battery replacement capabilities, Barrett et al. [134] took a different approach by using a mobile ground base with a robotic arm to replace the batteries of the drones. Alternatively, this design replaced the batteries through the top of the drone, whereas the others slid the batteries into the bottom side. Rather than the landing pad moving the drone, they decided to move the ground base to where the drone landed, and then replace the battery. All of these are viable solutions to keeping the drones in the air [134].

There are a number of ways to approach this problem: To start, the direction from which the battery will be inserted needs to be defined. It is possible that the battery could be inserted from the top, side, or bottom. If the battery is loaded from the top, it ultimately correlates to a robotic arm inserting that battery, which is not the most scalable design. Therefore, it would be ideal to load the battery from either the bottom or the side. Next, a method for moving the drone to the center of the landing pad is required. As discussed by Fujii et al. [132], a landing platform was used with two L-shaped arms with a fourbar linkage (see Figure 51a). With servomotors actuating the four bars, the two L shapes were forced to the center and, respectively, centered the drone. Although this seems to be a good approach, it is missing one key factor: it does not rotate the drone. If a drone is coming in for a landing and some unpredictable event occurs and it lands rotated an arbitrary direction away from the battery swapping mechanism, it could fail to swap out the batteries. Furthermore, this method is not fit for correcting the rotation of the drone. A more effective method has been developed to both center the drone and rotate it to the correct position so that the battery can be removed.

Suzuki et al. [135] used magnetic connections to cause contact between the battery and drone, whereas in the study by Herath et al. [136], they applied passive connections for power and cell balancing. Herath et al. [136], among others, used a simple method of centering and securing the drone, before replacing the battery. The designs included conical cavities for passively guiding the drone into the proper location. The tactic shown below has proven to be effective, but it lacks the compensation for different sized drones (see Figure 51b). Suzuki et al. and Herath et al. [136] used different locking mechanisms for securing the battery, but both of the groups loaded the battery into the drone from the bottom. A rack and pinion mechanism was used to push the battery onto a platform that elevated, inserting the battery into a mechanical coupler. Alternatively, the method proposed allowed the battery to slide horizontally through the drone [136,137].



Figure 51. View of (**a**) self-centering landing platform for drone [132], (**b**) passive landing guides, and (**c**) passive connectors for power and cell balancing [136].

Wireless charging provides an alternative to wired charging. Depending on the type of wireless charger, oriented landings may or may not be necessary [138]. A slight disadvantage of wireless charging is that the drone/UGV will have to be equipped with receiver coils to be able to charge (see Figure 52).



Figure 52. View of (**a**) a landing pad that uses wireless charging coils to charge quadcopters with receiver coils [139], (**b**) SKYCHARGE drone wireless charging pad [140], (**c**) WiBotic drone-charging pad [141], and (**d**) WiBotic UGV charging pad [141].

3.2. UGVs Deployed by Drones

UGVs can be deployed by drones in an integrated manner. Having a UGV be deployed by a drone is advantageous for several reasons. One of the main advantages is that the drone can fly the UGV over terrain that would have been impossible to navigate by the UGV. Another advantage is that the drone can also be used to fly the UGV directly to and from the mission area, which can save time and shorten the UGV's driving time. One potential disadvantage is that the UGV's size and weight are limited by the drone's carrying capacity.

Galeone developed an autonomous drone that is capable of transporting a ground robot through a gripping system (see Figure 53a) [142]. Hament and Oh developed a technical design requirements for a team of autonomous drones and UGVs for civil infrastructure inspection (see Figure 53b) [143]. In 2022, the Heven Drones Company designed and built a heavy lift drone that can carry a Micro Tactical Ground Robot (MTGR) (see Figure 53c) [144]. In 2022, a group in China tested a drone deploying a quadrupedal unmanned ground vehicle (see Figure 53d) [145]. The Korea Atomic Energy Research Institute used a combination of a drone and a snake robot for accident monitoring (see Figure 53e) [146]. Kawasaki's K-Racer tested an autonomous helicopter, which could carry a cargo robot (see Figure 53f) [147]. Nagatani et al. developed a robotic system for volcano exploration using drones transporting ground robots (see Figure 53g) [148].



Figure 53. View of (**a**) drone transporting a ground robot [142], (**b**) team of autonomous drones and UGVs for civil infrastructure inspection [143], (**c**) a drone that can carry a Micro Tactical Ground Robot (MTGR) [144], (**d**) drone deploying a quadrupedal unmanned ground vehicle [145], (**e**) a drone that can carry a snake robot for accident monitoring [146], (**f**) autonomous helicopter which transport a cargo robot [147], and (**g**) a robotic systems for volcano exploration [148].

4. Mining Challenges for Multi-Agent Systems

The environmental conditions in an underground mine present some challenges to the humans and machinery that work there, making their deployment difficult, if not impossible altogether. Studies into the use of autonomous systems have made it obvious that the proper utilization of these systems can significantly improve the search and rescue outcomes of underground mining emergencies. The intent for their deployment is generally to hasten the rescue mission process because, in an emergency, time is of the essence. They also help to reach inaccessible parts of the mine relatively easier and identify trapped miners quicker. It is no doubt that these systems are going to be faced with performanceimpeding conditions [4]. Addressed in this section are the challenges the robots are going to face.

4.1. Darkness

Since the inception of underground mining activities, illumination has been pivotal to the safety and performance of both miners and machinery. Every country has specifications for lighting intensity that are used in their mining industry (see Table 1).

Table 1. Illumination levels around machinery in the mining industries in different countries [149].

Country	Illuminance (Lux)
Belgium	25
Hungry	20–30
Canada (British Columbia)	53
Poland	10
Germany	80
United States of America	0.061 Fl/021 (cd/m ²)

The United States is the only country that opted for the measure of luminance over illuminance, as it is more correlated to what the eye perceives [150]. The lighting conditions in an underground mine, and especially, in an underground coal mine, are affected by a number of factors including (but not limited to) the following:

Poorly reflective walls: Reflectivity is a measure of the ability of a surface to reflect the incident light supplied to it. In lighting applications, surfaces or walls reflect a relatively high percentage of light, which helps in providing good visibility of the objects. In underground coal mines, exposed surfaces, and walls, especially around the ribs and faces, have very low reflectance, usually 1–10%, with 4% being the typical value [150]. Coupled with the low reflective nature of the environment is the texture-less and locally similar structure [151]. This presents localizability issues and severe sensor degradation. To address this issue, Mascarich et al. designed a multi-modal unit that incorporates a visible light camera with inertial sensors, as well as bright LEDs that flash only when the shutter opens [152]. When the multi-modal unit is used together with a visual-inertial odometry pipeline, it enables reliable robot navigation in the darkness. Additionally, other researchers have developed a system that integrates the robot's multi-modal system with a set of associated fusion algorithms that enable Simultaneous Localization and Mapping (SLAM) and the mapping of visually degraded environments in an underground mine [151].

Strict regulations surrounding the use of electrical equipment: Provision 75.1719–1(d) of the Codes of Federal Regulations (Mine Safety & Health Administration, 1988 [153]) states that: "The luminous intensity (surface brightness) of surfaces that are in a miner's normal field of vision of areas in working places that are required to be lighted shall not be less than 0.2 cd/m2 (0.6fl)." Additionally, for underground mines, the voltage of the lighting system should not exceed 125 volts. The electrical equipment must be intrinsically safe, which means it has to be safe for use in atmospheres containing flammable gasses and combustible dust [154]. Therefore, it implies that the electrical device should be incapable of producing enough thermal or electrical energy to ignite the fuel and cause an explosion. These regulations are in place to mitigate accidents and disasters, and rightly so,

underground mines make it impossible to use high luminescence output lamps in underground mines. For this reason, more research is required to know the feasibility of using high-output or low-voltage lighting systems.

4.2. Dust and Smoke

Underground mines can be broadly categorized into two main types: non-coal and coal ones. The non-coal mines are generally more metalliferous and deep-seated ones relative to the coal mines. Underground non-coal mines are considered to be relatively less dangerous than coal mines are because there are generally no poisonous emissions in those mines [155]. Underground coal mines are surfeited with noxious gasses such as carbon monoxide and firedamp (methane–air mixture), which can explode with the application of the slightest amount of heat and produce thick clouds of dust and smoke. The occurrence of spontaneous combustion in coal mines and the presence of dust and smoke can render the robots ineffective by directly impeding their capacity to self-localize and locate trapped miners with any accuracy [156].

Underground mine exploration activities produce heavy dust and smoke that obscures distinctive features and landmarks that the system may rely on for localization. According to [156], visual-inertial SLAM methods are unstable and can lose track or even enter unrecoverable states in such environments. The research on overcoming this issue is widely ongoing [157]. Reliance on LiDAR sensing modalities is the best available option, though it is insufficient. However, recent research focuses on multi-modal sensor fusion and robust state estimations to provide reliable alternatives [158].

4.3. GPS-Denied Environements

The implementation of UGV and drones is becoming an increasingly efficient tool for rescue applications, site and infrastructure inspection, remote controlling, exploration, path finding, and risk recognition [18,25,29,35,38]. During an underground emergency, the location of the base points plays a fundamental role in the rescue mission, and the hybrid robotic system can help to locate trapped miners easily. Ideally, the UGV and drones would be equipped with a Global Positioning System (GPS) to enable it to self-localize and map out the underground mine; however, the depth of the mine below the ground level, the rock composition, and the harsh environment in an underground mine make it difficult to maintain any usable GPS signal [4].

To mitigate this problem, robots deployed for GPS-deprived applications implement alternative positioning algorithms, such as Simultaneous Localization and Mapping (SLAM), Partially Observable Markov Decision Process (POMDP), Dijkstra, or Model Predictive Control (MPC), to mention a few [4–6]. These algorithms can help the robots to self-localize by using measurements from features inside the mine to track the occluded targets in GPSdenied atmospheres and map the morphed underground tunnels. The parameters in these algorithms will depend on the location, motion, and speed of the unmanned vehicle performing in the underground environment [4–6,151].

According to MSHA, these base points or targets are the keys to streamlining the rescues, finding the miners in an emergency, and keeping the rescue team safe [153]. For that reason, the GPS-denied environment is one of the most crucial challenges the robot can face in underground mines. [159] The location of strategic points, such as a fresh air base, allows the rescue teams to communicate between the command center and plan strategies [159,160].

Another strategic point is the refuge chamber: it isolates the miners from the hazardous event and has tools for establishing communication with the command center and rescue team to notify the situation in the mine in real time [161,162].

4.4. Dynamic and Harsh Environment

In the last two decades, improvements to the regulations and technological applications have helped to minimize accidents in underground mines. The fatalities and financial losses have decreased considerably compared with those in the last century [155,159]; however, the inherently harsh and dynamic nature of the underground environment continues to present the mining industry with a lot of challenges [157]. The harsh environment refers to the factors that potentially lead to a degraded perception of miners, difficulty breathing, inaccessibility to certain areas, and unstable communication signals in rescue practices and mining work. Some of these factors are dust, smoke, darkness, toxic gasses, unpassable obstacles, temperature, and humidity. These conditions are characterized by constant and sometimes sudden changes due to mining activity or progressing through the ore body, making an underground mine a dynamic atmosphere [154].

In 2006 in Sago Mine (coal mine), eleven out of twelve miners died of carbon monoxide poisoning before the rescuers reached them 41 h after a disaster [154]. The Sago Mine incident exemplifies the dynamic spaces and harsh environment that the miners and rescue teams must consider during incidents [154,159]. The occurrence of an explosion could cause the roof to collapse, and this can produce dense dust and debris. The debris or rubble can end up blocking some spaces, creating blind spots; this makes the exploration process complex [158]. These obstructed areas serve as toxic gas concentration sites because air cannot circulate through them. Additionally, the presence of gasses such as methane and oxygen can cause another explosion, and the cycle may be repeated. These conditions tend to slow down the already slow process of mine rescue, as the safety of the rescue team can also be endangered if these factors are ignored. The use of the UGV–drone hybrid mitigates the risk that rescuers would otherwise face in the mine because they can traverse toxic environments, reach obstructed spaces, and see in the darkness. This helps to significantly accelerate the rescue protocols and procedures [154,155,162].

The technology integrated into the UGV and drone monitors accurate results of the conditions under challenging environments and determines the occurring events in the mine. Technology such as RGB-depth (RGB-D) cameras measure the distance between obstacles. The hybrid UGV–drone system can reach more search areas; when the UGV cannot advance due to an obstruction, the drone will continue the exploration by air [162]. Similarly, the thermal camera can identify fire and analyze the robot's risk conditions during rescue procedures.

4.5. Ventilation and Air Velocity

Mine ventilation is the process of supplying sufficient fresh air to the underground environment through a ventilation system and working to ensure its proper distribution and use using controls that return contaminated air to the surface [163]. This fresh air volume must be sufficient to dilute and remove dust and noxious gasses, typically SO_2 , CH_4 , CO_2 , and CO, and regulate the temperature inside the mine [155,159,164]. The ventilation system is closely related to the development of work and rescue procedures. Confined spaces do not allow efficient air to flow through the mine; this makes the underground mine a place where harmful gasses and pollutants easily concentrate, with the least amount of deficiency in the ventilation system [158,165]. Fresh air enters the mine from the surface via the hoisting shafts and is distributed strategically using a series of fans and the space circulation (raises and ramps) of the mine to maintain the airflow [159,166].

The most important design parameter in monitoring the functionality of the ventilation system is air velocity; this parameter plays an important role in health and safety hazard management [159,164]. The ventilation system may have a stable flow, allowing the person to identify critical locations in real time; the air velocity can be measured using different sensors or techniques such as a rotating vane anemometer, a velometer airflow measure probe, a piezoresistive sensor, and an ultrasonic flowmeter [165,167]. During a disaster, however, the situation may be quite different. Fires or explosions may release dangerous gasses into the atmosphere. A disrupted ventilation system could result in an oxygen-deficient atmosphere and/or a buildup of toxic or explosive gasses [154,159]. The flow's velocity and direction permit the identification of critical locations where the gasses are concentrated at and measures them using technology integrated into the UGV and drone, such as thermal cameras, LiDAR, sensors, and others.

According to Mine Rescue Team Training in metal and non-metal mines, gas detection is essential to any rescue or recovery operation. The rescue teams must perform frequent tests for gasses as it advances beyond the fresh air base, identifying what harmful gasses are present, how much oxygen is in the atmosphere, and whether the gas levels are within the explosive range [152,159]. The sensors integrated into the UGV–drone system could monitor the gas concentrations, such as methane from 0.0 to 100 percent of volume, oxygen from 0.0 percent to at least 20 percent of volume, and carbon monoxide from 0.0 parts per million to at least 9,999 parts per million (30 CFR 49.6(a)(6)) [164].

The Fresh Air Base (FAB) is a structure designed as a temporary stop during an emergency by the miners and the rescue team [159]. The base is used to plan, rest, move forward and execute the evacuation/rescue operations [168]. It is equipped with a supply of potable water and compressed air, isolating the toxic gasses and giving rescue teams a breath of fresh air. Additionally, the base usually has a copy of the most recent emergency procedures, the map of the mine, and the communication route to the surface [167]. The ventilation engineers design the fresh air base based on the fresh air source, and it is connected to the ventilation system to guarantee breathable air across the mine. The appropriate use of the base permits them to establish communication between the rescue team and the UGV– drone system, increasing the communication range, guaranteeing the control of the robot, and making the signal more stable [169,170].

The rescue personnel can control the drones in the fresh air base, increasing the coverage of the UGV and drone, protecting the rescue team from hazards present in the disaster, and providing fresh air or breathing apparatus with oxygen. The control of a robot in a safe place reduces the risk posed to the rescuers and speeds up the rescue procedures. "A team's time underground is usually limited to two hours or less. The exact amount of time is determined both by the underground conditions and the type of apparatus being used. The distance you can advance also depends on underground conditions." [158,159].

4.6. Muddy Terrain

The rescue and self-escape missions are heavily time-dependent missions. The faster that these missions are executed, the higher the survival chance of the trapped miners is. One of the reasons that UGVs and drones have been deployed is to reduce the mission's execution time. The floor of the gallery is typically harsh and uneven with protuberances and even cracks on its surface. This condition does not hinder the performance of the machinery underground because they have large tires or tracks. In the event of an emergency, such as a roof fall, these roadways become filled with piles of rubble. Additionally, in the case of an inundation, heavy dust settling in water can form a thick slurry or muddy waters. The unevenness of the roadway, the access paths being filled with rubble, and the presence of muddy terrain can impede the movement of the robots [151]. Sandia National Laboratory, recognizing this challenge, developed the "Gemini Scout", shown in Figure 54, which overcame these movement-restricting factors [171].

The Gemini scout UGV is equipped with gas sensors, thermal cameras, and a tall, mounted tilt camera. At the same time, it is lightweight enough to crawl over boulders, rubble piles, and standing or flowing water. Its articulated body and rubber tracks enable it to maneuver over rough terrains.



Figure 54. View of Gemini Scout UGV [171].

4.7. Permissibility

Permissibility refers to the consent by the Mine Safety and Health Administration (MSHA) to allow electric face equipment used for mining operations, including mining rescue, in the mine [37]. The certification of approval is one of the barriers to the electronic function of the UGV and the drone, especially in coal mines. Permissibility is a major barrier to implementing such technologies.

According to MSHA (2012) in the 30 CFR § 75.506, "Electric face equipment", requirements for permissibility related to "electric-driven mine equipment and accessories manufactured on or after 30 March 1973, will be permissible electric face equipment only (1) if they are fabricated, assembled, or built under an approval, or any extension thereof, issued by the Bureau of Mines or the Mine Safety and Health Administration in accordance with schedule 2G, or any subsequent Bureau of Mines schedule promulgated by the Secretary after 30 March 1970, which amends, modifies, or supersedes the permissibility requirements of schedule 2G, and (2) if they are maintained in a permissible condition." [27,37].

The MSHA is the regulatory body in charge of granting permissibility certifications and practicing tests to guarantee the health and safety of mines [155]. The high concentrations of gasses amidst electrical equipment can generate an explosion due to the ignition of electrical devices incorporated in the robot [154,159,170]. However, the application of underground mining technologies, at least in the case of the hybrid ones, helps with the enhancement of rescue procedures, minimizing the manpower, and restoring reliability in mining operations. Therefore, the industry continues to investigate the application of permissive gadget technology in subway mines to maintain the safety of miners and rescue teams [27,154]. In addition, the multi-agent system implemented through UGVs and drones maintains the permissive requirements in terms of software design. The software has the capability to become an intrinsically safe system. Human factors and inadequate procedures are major contributors to fatalities in emergency events, and the area of autonomy and navigation for mine emergency applications will significantly improve health and safety.

4.8. Wireless Communication

A critical aspect of a multi-agent hybrid robotic system is the efficient and secure communication between the robotic agents necessary for transmitting the data. Similarly to the infeasibility of deploying GPS technology in subterranean environments, significant challenges arise for all forms of wired or wireless communication due to the lack of the infrastructure or the impracticality of installing such infrastructure necessary for communicating the information-carrying signal. Therefore, the design and deployment of an MAHRS must address this lack of communication tools by incorporating appropriate technologies and methodologies that enable a capability inherent to MAHRS to create and manage its own communication network.

Vehicle-to-Vehicle (V2V) communication technologies could provide the means to achieve efficient inter-communication between the robotic agents of MAHRSs. V2V communication has attracted the interest of considerable research in the field of Intelligent Transportation Systems (ITS) in recent years. Although, V2V is mainly studied in ITS applications, with the main purpose of enhancing road safety and driver protection (through integrating the speed and orientation of one vehicle with the collision zones of the surrounding vehicles), the same technologies can be applied for enabling efficient data communication between the members of MAHRSs. V2V is primarily achieved by employing Vehicular Ad Hoc Networks (VANETs) between vehicles, i.e., spontaneously created wireless networks of vehicles, in order to exchange data between two moving autonomous vehicles. These data can include collected sensor data, vehicle size, position, speed, heading, lateral/longitudinal acceleration, yaw rate, throttle position, brake status, steering angle, wiper status, turn signal status, and loss of stability information.

VANETs can be built by using any wireless networking technology. The most common ones are short range radio technologies such as WLAN, and especially, the IEEE 802.11p WLAN protocol, which was built with the intention to be used for vehicular networks. Cellular networks are also a promising candidate for the basis of making a VANET. Infrared transmission and reception can also be used for creating these wireless networks by using Visible Light Communication [172].

Araniti et al. examined the capabilities of using LTE for supporting V2V communications, as well as the advantages of LTE over the 802.11p [173]. They conclude that the strengths of LTE (high capacity, wide coverage, and high penetration) can cope with the drawbacks of 802.11p (poor scalability, low capacity, and intermittent connectivity). Visible Light Communication (VLC) based on LEDs for transmitting data is another applicable methodology. In environments were autonomous agents have to operate under smoke, dust, or poor illumination conditions, which deteriorate the onboard sensors accuracy and increase the risk of collisions, LED-based V2V technologies could offer promising solutions [174].

Besides VANETs, flying ad hoc networks (FANETs) have attracted a lot of interest in the last two decades. Drones, due to their high maneuverability and scalability, have been the most popular candidates to create autonomous ad hoc networks [175–177]. As a result, there is an abundance of literature sources that rigorously investigate multiple aspects of designing and implementing swarms of drones, which include (but are not limited to): swarm coordination [178–180], routing [181], security [182–187], and energy considerations [188], as well as their challenges and prospective uses [189].

4.9. Power Sources: Energy for Multi-Agent Systems

In underground environments such as coal mines, fuel- and electrically powered machines pose two principal problems: they are a potential fire and explosion hazard and produce noxious exhaust gasses that must be diluted to safe levels. Manufacturers of these machines incorporate special design features that aim to produce the least amount of heat and noxious gasses. The MSHA approval and certification center evaluates the power package to ensure that the special design features provide the protection that is required, and if federal requirements have been met, a certification is issued. Concerning the use of multiagent systems underground, all forms of power supplied to this system and any electrical systems installed on it must incorporate permissibility protection provided by the certified power package.

4.9.1. Batteries

The most important feature of a UGV is mobility. The power supply's duration is undoubtedly one of the most important indicators for evaluating a UGV, aside from the working temperature range. According to MSHA, battery-powered equipment should have an intrinsically safe active voltage circuit, which is a circuit that utilizes non-passive components such as transistors or other solid-state devices that either shunt overvoltage or overcurrent conditions or limit the output of the power source fast enough to prevent the ignition of a methane–air atmosphere [190]. The most common power supply source for UGVs is a battery, especially for small- and medium-sized UGVs. Batteries have some advantages over combustion engines: they are quiet, mechanically simple, and, most importantly, have lower working temperature ranges and no heated exhaust. This is preferable, especially for use in environments with methane–air concentrations.

Battery Type and Permissibility

UGVs, drones, and robotic platforms are commonly powered by batteries. The weight, size, and endurance of the battery are limitations of the performance of many autonomous vehicles [191]. In underground rescue applications, unmanned systems need a reliable backup power system to operate during prolonged recovery procedures, and it is a continuous challenge to select the best batteries that will perform safely [192]. The voltage level, capacity, discharge rate, activation time, charging time, lifespan, and their cost efficiency are some of the parameters that must be considered when one is selecting a battery pack for unmanaged systems. Autonomous vehicles have an integrated technology that operates with different functions. That means the battery loads and power demands vary depending on the operation requirements carried out in underground mines.

Lithium-based batteries are the most used technology in unmanned systems due to their higher energy density and charge–recharge efficiency [193]. Mostly, the unmanaged system industry uses lithium polymer (Li-Po) and lithium–ion (Li-ion) batteries to develop complete systems in autonomous vehicles. Furthermore, they are developing technology using lithium sulfur batteries (Li-S), where the energy density is even higher compared with that of other Lithium-based batteries [194]. Additionally, the old technology implemented in unmanaged systems was nickel-base batteries, but their usage has decreased in frequency due to the power-to-weight ratios for these types of batteries in drone engines not being efficient for the rate of climb and power consumption [195].

According to the National Institute for Occupational Safety and Health (NIOSH), the uses of Li ion batteries in some devices employed in mining are permissible for communication and tracking devices. Nevertheless, the Li ion battery cells pose a CH4 explosion hazard from a cell internal short circuit. The use of this type of battery requires further testing to rule out its use and permissibility in underground [194] mines. The sealed enclosure for batteries is one of the keys means of avoiding the ignition of explosions or fires in underground mines. On the other hand, an MSHA-approved Li ion cap lamp battery contains a multicell battery pack within a sealed plastic enclosure [192]. The batteries used lithium ion technology, which requires special consideration and handling techniques due to the extremely high energy density [196]. Quality control tests for permissible Li ion and Li polymer batteries are carried out to standardize their use in coal mining equipment, including UGVs and drones [196].

4.9.2. Fuel Gas

Secured operations of diesel-powered machinery in underground coal mines, where methane and other flammable gasses may be present, require custom features to mitigate fire and explosions. These features are listed in part 36, Title 30 of the Code of Federal Regulations (30 CFR 36). A permissibility checklist has been developed to recognize the diverse nature of the techniques used by equipment manufacturers to provide the required protection and the importance of ensuring that the safe devices continue to provide the required protection. If a diesel-powered engine system is going to be incorporated into a multi-agent system, then the engine and exhaust system surfaces must be maintained below $302 \otimes [197]$. One of the most effective means for reliably controlling this temperature has proven to be water jacketing the exhaust manifold and the associated piping. If the temperature around the engine goes beyond the above-mentioned limit, combustible

concentrations of methane surrounding the engine may be ignited by flames passing back through the engine intake system, passing through the joints in the engine intake or exhaust system.

4.9.3. Liquid Hydrogen

Liquid hydrogen is one of the most common fuels used in fuel cells. The supply of hydrogen is incomparable to that of conventional energy sources such as diesel, gasoline, or electricity due to their production scale. Additionally, hydrogen storage is the most serious limiting factor to its effectiveness and distribution. Despite the fact that hydrogen is as clean as the energy that is used to produce it is, liquid hydrogen in UGVs must be compressed in expensive high-pressure tanks, which requires more energy. Regardless of all of the disadvantages of using hydrogen fuel, hydrogen cell systems can outperform battery systems by a factor of 2–8 (within the 100–1500 Watt power range) [198]. This translates to a longer performance time and a better performance of the electrical components.

5. Application of Hybrid Systems in Underground Spaces

Multiple researchers have focused on integrating different sensors into a navigation system that is capable of exploring and collecting data in subterranean or similar GPS-deprived spaces, such as catacombs, caves, and mines.

Although there are innumerable sources on the international literature showcasing the development of UGVs deployed in a GPS-denied environment, both surface and underground applications, an exhaustive review is omitted for the sake of simplicity. However, some examples of autonomous UGVs deployed in the mining industry can be insightful regarding the vast possible capabilities of such vehicles: Lee et al. deployed the Cave Crawler platform equipped with a spinning laser scanner (i.e., LiDAR), an IMU (Inertial Measurement Unit), and wheel encoders in an underground coal mine (see Figure 55a). A band-based ICP algorithm (Iterative Closest Point) was developed for reconstructing a narrow band from the 3D point cloud of the mine tunnels to reduce the computational cost [199]. Nagatani et al. integrated a 2D LiDAR scanner with an infrared camera with a thermal ICP algorithm to develop 3D thermography maps, which help to locate humans (see Figure 55b) [200]. Lavigne and Marshall integrated a laser scanner and an IMU with an RFID reader (Radio Frequency Identification) that detects passive RFID beacons for a landmark-based method for mitigating drift in large-scale underground mine mapping (see Figure 55c) [201]. Neumann et al. implemented a SLAM algorithm using ROS packages (Robotic Operating System) to fuse the data from LiDARs, IMU, and wheel encoders into a collision avoidance navigation framework for underground mines (see Figure 55d) [202].

Similarly, there is an abundance of sources in the international literature that showcase the development and the capabilities of drones in a plethora of environments [3,4,91,203,204]. The deployment of drones in mines or similar GPS-deprived spaces has attracted a fair share of research interest, especially in recent years, which have advanced localization and mapping methodologies (e.g., SLAM), as well as computational power and efficiency. Shahmoradi et al. provided a comprehensive review of the deployment of drones in the mining industry, with applications that range from 3D mapping and geotechnical characterization to mine operation monitoring, air quality monitoring, subsidence monitoring, and mine rescue missions [4]. Jones et al. showcased a multitude of similar implementations of drone technology in underground mining operations, focusing on the Hovermap drone system that was developed by the Australian Commonwealth Scientific and Industrial Research Organization (CSIRO) for GPS-deprived environments [205].

The recent advances in computing, sensors, hardware, and software have led to an increasing interest in deploying cooperative UGV–UAV systems in GPS-denied environments. Asadi et al. showed that researchers have been exploring the capabilities of multi-agent UGV–UAV systems since the 1990s for different environments and applications. The reviewed applications are collectively summarized in a tabular format in a very insightful

way [91]. However, the same study concludes that most of the reviewed applications do not operate the collaborative robots simultaneously or autonomously. More relevant research on the deployment of multi-agent systems in underground, GPS-deprived spaces was conducted through the 'DARPA Subterranean Challenge' (Defense Advanced Research Projects Agency). This challenge brought together researchers and innovators from around the world to develop systems that can map, navigate, and execute time-critical tasks in a complex subterranean environment that combines tunnel systems (i.e., underground mines), urban underground layouts, and natural cave networks. The vast variety of the multi-agent systems that competed in the challenge included (but were not limited to): the winning team CERBERUS, who combined walking and flying robots [206], the team CSIRO Data61, who deployed multiple types of ground and air robots: a hexapod, a quadruped robot, a UAV, and three tracked robots [207], the CTU-CRAS-NORLAB team, who deployed wheeled, tracked and crawling (hexapod) ground robots and quadrotor aerial robots [208], while the NeBula team combined wheeled, tracked, quadruped, aerial (quadrotor and coax-octocopter), as well as hybrid robots [209].



Figure 55. View of (**a**) the Cave Crawler platform with a spinning laser scanner in an underground coal mine [199], (**b**) Kenaf robot with a 3D laser scanner and an infrared camera [200], (**c**) the custom hardware enclosure of the landmark-based mapping platform with a laser scanner, an IMU, and a RFID reader [201], and (**d**) the collision avoidance mobile testing platform [202].

6. Conclusions

This review focuses primarily on unmanned ground and aerial vehicles, either as stand-alone agents or as multi-agent systems (hybrid) in underground spaces. The categories of unmanned surface and underwater vehicles (USVs and UUVs), which encompass autonomous agents that operate on the surface and under the surface of the water, respectively, are not discussed in this review. The principal reason for this omission is that these types of unmanned vehicles have little or no applicability in underground environments or can be easily replaced by hybrid UGV–UAV systems. In addition to the examination of the various configurations of ground vehicles and drones prevalent in the global literature, the category of bio-inspired robots is briefly examined. The emphasis of the examination

44 of 54

of these robotic agents is on their efficiency and capacity to carry out, either as stand-alone agents or as part of an MAHRS, conducting exploratory and surveying missions in subterranean spaces.

The literature shows an abundance of different configurations of ground vehicles (UGVs) and drones (UAVs), as well as coupled or inter-transformable UGV–UAV systems that have the capacity for autonomous navigation, and hence, the exploration of subterranean spaces. A similar abundance of cooperative combinations of these unmanned vehicles can be found in the literature. These multi-agent hybrid robotic systems (MAHRSs) are meant to be as terrain versatile as possible by using specific unmanned robots to support and make up for the shortcomings of the other cooperative agents. A major advantage of MAHRSs is the efficiency, in terms of speed and deployment time, that they can contribute in a mission. Such systems assign separate subtasks to each of their robotic agents according to their built-in capabilities. In this way, the ultimate mission of the system is executed more speedily and efficiently.

The principal aspects that a robot's efficiency is commonly measured against include (but are not restricted to) the following: size, energy capacity/consumption, and movement control. Each of these parameters determines some capabilities, as well as the effectiveness of them. At the same time, these parameters are not independent of each other, but instead they affect and are affected by each other. Although a detailed discussion of these aspects is out of the scope of this study, the following indicative observations can be made. Size and weight are important considerations for any hybrid system that will be used within an underground mine. UGVs that deploy drones will both have increased sizes and weights compared to those of a UGV deployed via a drone because they must carry the apparatus to deploy and house the drones. The size of a robot determines the feasible deployment working environments, the overload capacity, and its maneuverability. The overload capacity of a robot determines whether it can house and deploy other agents or can be deployed by other agents. Moreover, the overload capacity directly determines the onboard hardware, and hence, the computational power and overall functionality of the robot. Naturally, the size of a robot affects its battery consumption and maneuverability. The energy capacity/consumption relates directly to the deployment time, onboard hardware, computational cost, and actuation mechanisms. In other words, heavier and smarter robots usually require more energy, but they can accomplish more complex tasks. In general, the sizes of underground mines necessitate that any deployed system should have a long endurance capacity. UGVs that deploy tethered or untethered drones will consume a significant amount of power due to their large size and the need to power or charge the deployed drones; however, they can compensate for this by carrying a large battery that can give them a long endurance capacity. A drone that deploys a UGV would consume a lot of power because it has to fly the UGV to its destination; this would reduce the endurance of the system. The deployed UGV would also have a limited battery size and weight because it has to be light enough to be picked up by the drone; this would limit its endurance once it has been deployed. Finally, movement control determines the feasible deployment working environment of the robot, whereas it affects energy consumption and is interrelated to the size. The complexity that arises from the interconnections and codependency of such design aspects is evident even when we are considering a few basic aspects. Therefore, the reasons underlying the popularity of designing MAHRS becomes obvious.

Hybrid systems consisting of unmanned aerial and ground vehicles can efficiently explore and navigate underground spaces such as catacombs, caves, and mines. The cooperative implementation of drones and ground-traversing vehicles means that they have a greater exploration capacity in dynamic underground environments that cannot be easily accessed from the ground. Naturally formed underground cavities such as caverns and lava tubes are usually inaccessible or even inhospitable for wheeled vehicles. Similarly, man-made underground openings, such as mines or tunnels, can be restrictive to rovers by construction or be rendered as such due to caving, infrastructure failures, or erosion. Exploring such underground spaces with a hybrid system means that multiple traversing alternatives can be crucial to exploration missions. Spaces which cannot be accessed by rovers or humans may be explorable by small, precise drones that can fly through tight spaces or reach high places to gather environmental data. These data can aid mine rescue teams or be used for surveying new environments.

Depending on the environment, UGVs deployed by drones could have the quickest deployment time because they can fly quickly to the deployment area, whereas UGVs that deploy either tethered or non-tethered drones must drive to their destination, which could take longer. Rovers deployed by drones are at a major disadvantage in underground mines because of the difficulty of flying in confined spaces while carrying the significant and cumbersome load of the rover. On the other side, tethered and non-tethered drones deployed by UGVs exhibit advantages in accessing underground mines because the UGVs can drive them to their deployment area. Subsequently, the deployed drones can utilize their relatively smaller size, which enables flight in confined spaces. Tethered drones deployed by UGVs can have a slight disadvantage depending on the intended range of the drone and the number of obstacles in the mine for the tether to overcome.

Bio-inspired robots can also be integrated in an MAHRS. The capabilities and efficiency of an MAHRS can be expanded or improved by such robots. Although, most applications in the literature showcase bio-inspired robots that are small in size, and hence, do not have the capacity to support advanced functionalities (with the exception of the legged robots such as the ones shown in Figure 10). Nevertheless, such small robots can be the ideal candidates for critical subtasks in the successful deployment of an MAHRS. Assigning the roles of wireless network nodes, distinctive landmarks, or signal beacons, or data stations to such robots would render vast flexibility to an MAHRS.

In any case, the robotic agents of an MAHRS that intend to be used for deployment in underground mines need to be carefully designed in order to overcome the challenges imposed by that particular environment: the poor illumination, as well as the presence of dust and smoke, impose limitations to the applicability of sensor modalities depending on the light conditions, while they can significantly occludes the data collected by other sensors; the GPS-denied nature of the subterranean spaces does not allow the use of one of the most powerful localization technologies; the dynamic and feature-less nature of the mines adds more complexity to autonomous navigation processes; ventilation and air velocity fluctuations create disturbances in the movement of UAVs, while uneven and muddy terrain can hinder the movement of a UGV; finally, permissibility regulations have to be considered in the design and fabrication of the robots. Nevertheless, multiple researchers have developed single, hybrid, or multi-agent systems that are optimized in some aspect for mines or similar subterranean environments, with insightful results.

To conclude, the ultimate scope of this study is to provide a collective literature review by drawing insights from Mineral Engineering R&D and Intelligent Robotic Systems R&D that aims to improve and expedite future R&D on intelligent underground systems in an effort to facilitate and inspire designers and innovators of such systems.

Author Contributions: Conceptualization, M.H., P.R., V.A., H.K. and S.S.; methodology, C.D. (Chris Dinelli), J.R., M.E., S.L., J.G., J.M., C.D. (Chase Dunaway) and M.H.; validation, V.A., H.K., S.S., P.R. and M.H.; formal analysis, C.D. (Chris Dinelli), J.R. and M.E.; investigation, C.D. (Chris Dinelli), J.R., M.E., V.A., H.K., S.S., P.R. and M.H.; resources, M.H. and P.R.; writing—original draft preparation, C.D. (Chris Dinelli), J.R., M.E., S.L., J.G., J.M., C.D. (Chase Dunaway) and M.H.; writing—review and editing, V.A., H.K., S.S., P.R. and M.H.; visualization, C.D. (Chris Dinelli), J.R. and M.E.; supervision, M.H. and P.R.; project administration, S.S., P.R. and M.H.; funding acquisition, M.H., P.R. and S.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by NIOSH-CDC: grant number U60 OH012351.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All data are available in the article.

Acknowledgments: This study was funded by the National Institute for Occupational Safety and Health (NIOSH) under the award #U60OH012351. The views, opinions, and recommendations expressed herein are solely those of the authors and do not necessarily reflect the views of NIOSH. Mentions of trade names, commercial products, or organizations does not imply endorsement by the authors nor the funding organization.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

Acronyms	Definitions
2D	Two-Dimensional
3D	Three-Dimensional
CFR	Code of Federal Regulations
CSIRO	(Australian) Commonwealth Scientific and Industrial Research Organization
DARPA	Defense Advanced Research Projects Agency
DOF	Degrees of Freedom
FAB	Fresh Air Base
FANET	Flying Ad hoc Network
GPS	Global Positioning System
ICP	Iterative Closest Point
IMU	Inertial Measurement Unit
ITS	Intelligent Transportation System
LED	Light Emitting Diode
LEMUR	Limbed Excursion Mechanical Utility Robot
LiDAR	Light Detection and Ranging
Li-ion	Lithium ion
Li-Po	Lithium Polymer
Li-S	Lithium–Sulfur
LTE	Long-Term Evolution
MAHRS	Multiple-Agent Hybrid Robotic System
MPC	Model Predictive Control
MTGR	Micro Tactical Ground Robot
MMC	Moving-Mass Controlled
MSHA	Mine Safety and Health Administration
POMDP	Partially Observable Markov Decision Process
R&D	Research and Development
RFID	Radio Frequency Identification
RGB	Red–Green–Blue
RGB-D	Red–Green–Blue–Depth
ROS	Robotic Operating System
SLAM	Simultaneous Localization and Mapping
SMA	Shape Memory Alloy
UAV	Unmanned Aerial Vehicle
UGV	Unmanned Ground Vehicle
USV	Unmanned Surface Vehicle
UUV	Unmanned Underwater Vehicle
V2V	Vehicle-to-Vehicle
VANET	Vehicular Ad Hoc Network
VTOL	Vertical Takeoff and Landing
VLC	Visible Light Communication
WLAN	Wireless Local Area Network

References

- Darvishpoor, S.; Roshanian, J.; Raissi, A.; Hassanalian, M. Classifications, configurations, and flight mechanisms of unmanned aerial systems: A review. *Prog. Aerosp. Sci.* 2021, 121, 100694. [CrossRef]
- Hassanalian, M.; Rice, D.; Abdelkefi, A. Evolution of space drones for planetary exploration: A review. Prog. Aerosp. Sci. 2018, 97, 61–105. [CrossRef]
- Hassanalian, M.; Abdelkefi, A. Classifications, applications, and design challenges of drones: A review. Prog. Aerosp. Sci. 2017, 91, 99–131. [CrossRef]
- Shahmoradi, J.; Talebi, E.; Roghanchi, P.; Hassanalian, M. A Comprehensive Review of Applications of Drone Technology in the Mining Industry. *Drones* 2020, 4, 34. [CrossRef]
- 5. Shahmoradi, J.; Roghanchi, P.; Hassanalian, M. Design, analysis and prototyping of a spherical drone for underground mines. *Int. J. Theor. Appl. Multiscale Mech.* **2022**, *4*, 58–82. [CrossRef]
- Mirzaeinia, A.; Shahmoradi, J.; Roghanchi, P.; Hassanalian, M. Autonomous Routing and Power Management of Drones in GPS-Denied Environments through Dijkstra Algorithm. In Proceedings of the 2019 AIAA Propulsion and Energy Conference, Indianapolis, Indiana, 19–22 August 2019.
- Shahmoradi, J.; Mirzaeinia, A.; Hassanalian, M.; Roghanchi, P. Monitoring of Inaccessible Areas in GPS-Denied Underground Mines Using a Fully Autonomous Encased Safety Inspection Drone. In Proceedings of the AIAA SciTech 2020, Orlando, FL, USA, 6–10 January 2020.
- Lanctot, S.; Cooke, J.; Montoya, J.; Herkenhoff, B.; Mostafanejad, A.; Hassanalian, M. A Hybrid Vehicle System for Depositing Payloads in Extreme Environments Like Antarctica and Arctic. In Proceedings of the AIAA Propulsion and Energy 2021 Forum, Virtual Event, 9–11 August 2021; p. 3726.
- 9. Louie, A.H. Relational biology of symbiosis. Axiomathes 2010, 20, 495–509. [CrossRef]
- Borges, P.; Peynot, T.; Liang, S.; Arain, B.; Wildie, M.; Minareci, M.; Lichman, S.; Samvedi, G.; Sa, I.; Hudson, N.; et al. A Survey on Terrain Traversability Analysis for Autonomous Ground Vehicles: Methods, Sensors, and Challenges. *Field Robot.* 2022, 2, 1567–1627. [CrossRef]
- Lanctot, S.; Herkenhoff, B.; Hassanalian, M. Unmanned Launching & Landing Rover (ULLR) for Moon and Martian Missions on Ice-Caps. In Proceedings of the 2021 ASCEND Conference, Las Vegas, NV, USA, 15–17 November 2021.
- 12. John, T.S. Advancements in robotics and its future uses. Int. J. Sci. Eng. Res. 2011, 2, 1–6.
- 13. Ben-Ari, M.; Mondada, F. Robotic motion and odometry. In *Elements of Robotics*; Springer: Cham, Switzerland, 2018; pp. 63–93.
- 14. Ni, J.; Hu, J.; Xiang, C. A review for design and dynamics control of unmanned ground vehicle. *Inst. Mech. Eng. Part D J. Automob. Eng.* 2021, 235, 1084–1100. [CrossRef]
- 15. Clearpath Robotics: Mobile Robots for Research & Development. Available online: https://clearpathrobotics.com/ (accessed on 30 December 2022).
- 16. Olmedo, N.A.; Barczyk, M.; Zhang, H.; Wilson, W.; Lipsett, M.G. A UGV-based modular robotic manipulator for soil sampling and terramechanics investigations. *J. Unmanned Veh. Syst.* **2020**, *8*, 364–381. [CrossRef]
- 17. Gonzalez-De-Santos, P.; Fernández, R.; Sepúlveda, D.; Navas, E.; Armada, M. Unmanned ground vehicles for smart farms. In Agronomy-Climate Change and Food Security; Intech UK: Horwich, UK, 2020; p. 73.
- 18. Blackburn, M.R.; Bailey, R.; Lytle, B. Improved mobility in a multi-degree-of-freedom unmanned ground vehicle. In *Unmanned Ground Vehicle Technology VI*; Society of Photo Optical: Orlando, FL, USA, 2004; Volume 5422, pp. 124–134.
- 19. Sun, B.; Jing, X. A tracked robot with novel bio-inspired passive "legs". Robot. Biomim. 2017, 4, 1–14. [CrossRef] [PubMed]
- 20. Guo, W.; Qiu, J.; Xu, X.; Wu, J. TALBOT: A Track-Leg Transformable Robot. Sensors 2022, 22, 1470. [CrossRef] [PubMed]
- 21. Herkenhoff, B.; Lanctot, S.; Bjorkman, T.; Serda, N.; Hassanalian, M. Preliminary Design Concept of Locust Inspired Jumping Moon Robot Swarm. In Proceedings of the AIAA Propulsion and Energy 2021 Forum, Virtual Event, 9–11 August 2021; p. 3270.
- Western, A.; Cervantes, R.; Dunning, C.; Haghshenas Jaryani, M.; Hassanalian, M. Bioinspired Robot with Walking, Rolling, and Jumping Capabilities for Planetary Exploration. In Proceedings of the AIAA Aviation 2021 Forum, Virtual Event, 2–6 August 2021; p. 2784.
- Herkenhoff, B.K.; Lanctot, S.I.; Fisher, J.M.; Serda, N.; Bjorkman, T.S.; Martinez, V.; Johnsonand, T.; Davis, C.; Yazzie, T.; Vadiee, N.; et al. Preliminary Design Concept of Locust Inspired Jumping Moon Robot Swarm. In Proceedings of the Lunar and Planetary Science Conference, Online, 15–19 March 2021; p. 2754.
- 24. Western, A.; Haghshenas-Jaryani, M.; Hassanalian, M. Golden wheel spider-inspired rolling robots for planetary exploration. *Acta Astronaut.* **2022**, 204, 34–48. [CrossRef]
- Rezazadeh, S.; Abate, A.; Hatton, R.L.; Hurst, J.W. Robot leg design: A constructive framework. *IEEE Access* 2018, 6, 54369–54387. [CrossRef]
- 26. Prada, E.; Miková, Ľ.; Surovec, R.; Kenderová, M. Complex kinematic model of snake-like robot with holonomic constraints. In Proceedings of the Mezinárodní Vědecká Konference k Problematice Technologických a Inovačních Procesů Technológia Europea, Hradec Králové, Czech Republic, 11–14 December 2012.
- Crespi, A.; Badertscher, A.; Guignard, A.; Ijspeert, A.J. AmphiBot I: An amphibious snake-like robot. *Robot. Auton. Syst.* 2005, 50, 163–175. [CrossRef]
- CDC—Mining Contract—Snake Robot for Mine Rescue—NIOSH. Available online: https://www.cdc.gov/niosh/mining/researc hprogram/contracts/contract200--2009-30721.html (accessed on 30 December 2022).

- Menciassi, A.; Gorini, S.; Pernorio, G.; Weiting, L.; Valvo, F.; Dario, P. Design, fabrication and performances of a biomimetic robotic earthworm. In Proceedings of the 2004 IEEE International Conference on Robotics and Biomimetics, Shenyang, China, 22–26 August 2004; pp. 274–278.
- 30. Carlo, M.; Metin, S. A biomimetic climbing robot based on the gecko. J. Bionic Eng. 2006, 3, 115–125.
- Saunders, A.; Goldman, D.I.; Full, R.J.; Buehler, M. The rise climbing robot: Body and leg design. In Proceedings of the Unmanned Systems Technology VIII, Kissimmee, FL, USA, 17–20 May 2006; pp. 401–413.
- Uckert, K.; Parness, A.; Chanover, N.; Eshelman, E.J.; Abcouwer, N.; Nash, J.; Detry, R.; Fuller, C.; Voelz, D.; Hull, R.; et al. Investigating habitability with an integrated rock-climbing robot and astrobiology instrument suite. *Astrobiology* 2020, 20, 1427–1449.
 [CrossRef]
- 33. The First Climbing Robot for Mars–Sciworthy. Available online: https://sciworthy.com/the-first-climbing-robot-for-mars/ (accessed on 30 December 2022).
- 34. Lanctot, S.; Montoya, J.; Dunaway, C.; Flores, C.E.; Barstow, J.; Janney, W.; Eisenberg, S.; Good, F.; Davis, N.S.; Zhang, S.; et al. Pillbug-Inspired Robot with Crawling and Rolling Locomotion Mechanisms for Use on the Lunar Surface. In Proceedings of the AIAA SciTech 2023 Forum, National Harbor, MD, USA & Online, 23–27 January 2023.
- 35. Liu, G.H.; Lin, H.Y.; Lin, H.Y.; Chen, S.T.; Lin, P.C. A bio-inspired hopping kangaroo robot with an active tail. *J. Bionic Eng.* 2014, 11, 541–555. [CrossRef]
- 36. Yu, J.; Ding, R.; Yang, Q.; Tan, M.; Wang, W.; Zhang, J. On a bio-inspired amphibious robot capable of multimodal motion. *IEEE/ASME Trans. Mechatron.* 2011, 17, 847–856. [CrossRef]
- 37. Han, S. A Guide to the Different Types of Drones & UAS. Everglades University. 5 March 2021. Available online: https://www. evergladesuniversity.edu/blog/guide-different-types-drones-unmanned-aerial-systems/ (accessed on 1 December 2022).
- Carholt, O.C.; Fresk, E.; Andrikopoulos, G.; Nikolakopoulos, G. Design, modelling and control of a single rotor UAV. In Proceedings of the 2016 24th Mediterranean Conference on Control and Automation (MED), Athens, Greece, 21–24 June 2016; pp. 840–845.
- Darvishpoor, S.; Jafar, R.; Morteza, T. A novel concept of VTOL bi-rotor UAV based on moving mass control. Aerosp. Sci. Technol. 2020, 107, 106238. [CrossRef]
- Mirkov, N.; Rasuo, B. Maneuverability of an UAV with coanda effect based lift production. In Proceedings of the 28th International Council of the Aeronautical Sciences (ICAS), Brisbane, Australia, 23–28 September 2012; pp. 1–6.
- Barlow, C.; Lewis, D.; Prior, S.D.; Odedra, S.; Erbil, M.A.; Karamanoglu, M.; Collins, R. Investigating the use of the Coanda Effect to create novel unmanned aerial vehicles. Microsoft Word–MES_2009_Paper_coanda effect C Barlow D Lewis (soton.ac.uk). In Proceedings of the International Conference on Manufacturing and Engineering Systems, Huwei, Taiwan, 17–19 December 2009; pp. 386–391.
- Lavars, N. Bizarre 'Bicopter' Drone Uses Two Tilting Rotors for 50-Minute Flights. Available online: https://newatlas.com/dro nes/v-coptr-falcon-drone-tilting-50/ (accessed on 20 December 2019).
- 43. Jin, H. Tricopter vs. Quadcopter 2022: Which Is Better for You. Available online: https://lucidcam.com/tricopter-vs-quadcopter/ (accessed on 27 April 2022).
- Yoo, D.W.; Oh, H.D.; Won, D.Y.; Tahk, M.J. Dynamic modeling and control system design for Tri-Rotor UAV. In Proceedings of the 2010 3rd International Symposium on Systems and Control in Aeronautics and Astronautics, Harbin, China, 8 June 2010; pp. 762–767.
- 45. Lukow, S.; Sherman, M.; Gammill, M.; Hassanalian, M. Design and Fabrication of Electromagnetic Attachment Mechanism for a Hybrid Drone for Mars Exploration. In Proceedings of the 2021 AIAA SciTech Forum, Virtual Event, 11–15 & 19–21 January 2021.
- 46. Pilot Institute. Is a Drone with More Rotors Better? Available online: https://pilotinstitute.com/more-rotors-better/ (accessed on 30 December 2022).
- 47. Lenski, P. Design, Construction and Operation of a Pentacopter. Master's Thesis, Luleå University of Technology, Department of Computer Science, Electrical and Space Engineering, Luleå, Sweden, March 2017.
- Foxtech. Blog–Foxtech Screamer Racing Pentacopter-Preorder. Available online: https://www.foxtechfpv.com/blog/0530-news/ (accessed on 30 December 2022).
- Guitarbob. Dragonfly/Pentacopter Project. Available online: https://www.instructables.com/DragonflyPentacopter-Project/ (accessed on 30 December 2022).
- 50. RedDrone. The Red Drone. (Pentacopter Version.right-Top View). Available online: https://www.123rf.com/photo_37276849_t he-red-dronepentacopter-versionrighttop-view.html (accessed on 30 December 2022).
- 51. Night Hawk, Sky Rider. King Soopers—Sky Rider Night Hawk Hexacopter Drone with Wi-Fi Camera—Blue/Black, 1 CT. Available online: https://www.kingsoopers.com/p/skyrider-night-hawk-hexacopter-drone-with-wi-fi-camera-blue-black/0004 732355700 (accessed on 30 December 2022).
- Octocopter. Spreading Wings S1000 DJI. DJI Official. Available online: https://www.dji.com/spreading-wings-s1000 (accessed on 30 December 2022).
- 53. Throneberry, G.; Hassanalian, M.; Hocut, C.M.; Abdelkefi, A. Insights on the potential of vibratory actuation mechanism for enhanced performance of flapping-wing drones. *Meccanica* **2021**, *56*, 1–16. [CrossRef]
- 54. Hassanalian, M.; Abdelkefi, A. Towards Improved Hybrid Actuation Mechanisms for Flapping Wing Micro Air Vehicles: Analytical and Experimental Investigations. *Drones* **2019**, *3*, 73. [CrossRef]

- 55. Throneberry, G.; Hassanalian, M.; Abdelkefi, A. Insights into Sensitivity of Wing Shape and Kinematic Parameters Relative to Aerodynamic Performance of Flapping Wing Nano Air Vehicles. *Drones* **2019**, *3*, 49. [CrossRef]
- Ghommem, M.; Hassanalian, M.; Al-Marzooqi, M.; Throneberry, G.; Abdelkefi, A. Sizing process, aerodynamic analysis, and experimental assessment of a biplane flapping wing nano air vehicle. *Inst. Mech. Eng. Part G. J. Aerosp. Eng.* 2019, 233, 0954410019852570. [CrossRef]
- 57. Hassanalian, M.; Abdelkefi, A. Methodologies for weight estimation of fixed and flapping wing micro air vehicles. *Meccanica* **2017**, *52*, 2047–2068. [CrossRef]
- Hassanalian, M.; Abdelkefi, A.; Wei, M.; Ziaei-Rad, S. A novel methodology for wing sizing of bio-inspired flapping wing micro air vehicles: Theory and prototype. *Acta Mech.* 2017, 228, 1097–1113. [CrossRef]
- Khan, M.M.; Gee, P.; Upshaw, J.; Hassanalian, M. Taxidermy Birds as Platform for Flapping Wing Drones: A Bioinspired Mechanism for Wildlife Monitoring. In Proceedings of the AIAA SciTech 2023 Forum, National Harbor, MD, USA & Online, 23–27 January 2023.
- Martinez-Ponce, J.; Urban, C.; Armanini, S.F.; Agarwal, R.K.; Hassanalian, M. Aerodynamic Analysis of V-Shaped Flight Formation of Flapping-Wing Drones: Analytical and Experimental Studies. In Proceedings of the AIAA SCITECH 2022 Forum, San Diego, CA, USA & Online, 3–7 January 2022.
- 61. Gammill, M.; Hassanalian, M. Aerodynamic Analysis of Manta Ray Inspired Fixed and Flapping-Wing Drones for High Altitude Venus Exploration. In Proceedings of the 2021 ASCEND Conference, Las Vegas, NV, USA, 15–17 November 2021.
- 62. Zint, N.; Silva, H.; Hassanalian, M. Design and Prototyping a Butterfly-Inspired Flapping-Wing MAV: Novel Applications of Drones for Met Gala Fashion Event. In Proceedings of the 2021 AIAA SciTech Forum, Virtual Event, 11–15 & 19–21 January 2021.
- 63. Maestas, S.; Martinez-Ponce, J.; Edwards, N.; Hassanalian, M. Applying Least Square Curve Fitting for Extraction the Birds' Wing Shapes: Flapping Wing MAVs and Kinematic Optimization. In Proceedings of the 2020 AIAA Aviation Forum, Virtual, 15–19 June 2020.
- 64. Erb, J.; Strauss, E.; Naghdi, M.; Hassanalian, M. Exploration of Venus' Upper Atmosphere Using an Aqua and Bacteria Inspired Aerial System. In Proceedings of the AIAA SCITECH 2022 Forum, San Diego, CA, USA & Online, 3–7 January 2022; p. 2581.
- Kawano, H. Study of path planning method for under-actuated blimp-type UAV in stochastic wind disturbance via augmentedmdp. In Proceedings of the 2011 IEEE/ASME International Conference on Advanced Intelligent Mechatronics (AIM), Budapest, Hungary, 4 July 2011; pp. 180–185.
- 66. Shrestha, E.; Hrishikeshavan, V.; Benedict, M.; Yeo, D.; Chopra, I. Development of control strategies and flight testing of a twincyclocopter in forward flight. In Proceedings of the 70th annual national forum of the American Helicopter Society, Montreal, QC, Canada, 20 May 2014.
- 67. Tank Quadcopter Drone Uncrate. Available online: https://uncrate.com/tank-quadcopter-drone/ (accessed on 30 December 2022).
- 68. Miniature Hybrid UAS/UGV Introduced Unmanned Systems Technology. Available online: https://www.unmannedsystemst echnology.com/2020/01/miniature-hybrid-uas-ugv-introduced/ (accessed on 30 December 2022).
- 69. Shapeshifting Pegasus Drone Both Drives and Flies, Now Has Smart Radio (forbes.com). Available online: https://www.forbes .com/sites/sebastienroblin/2021/01/27/shapeshifting-pegasus-drone-both-drives-and-flies-now-has-smart-radio/?sh=6a295650 795e (accessed on 30 December 2022).
- Available online: https://dronehub.ai/we-are-proud-to-present-the-huuver-our-new-hybrid-drone-that-flies-and-rides/ (accessed on 30 December 2022).
- Kim, K.; Spieler, P.; Lupu, E.S.; Ramezani, A.; Chung, S.J. A bipedal walking robot that can fly, slackline, and skateboard. *Sci. Robot.* 2021, *6*, eabf8136. [CrossRef]
- 72. Goldman, J. Parrot MiniDrone Rolling Spider Review: Inexpensive Drone Fun for Novice Flyers. Available online: https://www.cnet.com/reviews/parrot-minidrone-rolling-spider-review/ (accessed on 30 December 2022).
- 73. Mining & Technology: Drones to Map Underground Areas, Save Time & Improve Safety (mozambiqueminingpost.com). Available online: https://mozambiqueminingpost.com/2018/04/30/mining-technology-drones-to-map-underground-areas-save-tim e-improve-safety/ (accessed on 30 December 2022).
- 74. Ahmed, S.N. 3D scanning and mapping of underground mine workings using aerial drones. Phys. Can. 2018, 74, 103.
- 75. Biggs, T. Disney Research Creates Drone Car That Can Climb Walls. Available online: https://www.smh.com.au/technology/di sney-research-creates-drone-car-that-can-climb-walls-20151231-glx6dd.html (accessed on 30 December 2022).
- 76. Panther. Advanced Tactics Panther Drone Has Completed First Aerial Package Delivery Test. Available online: https://uasweekly.com/2017/03/31/advanced-tactics-panther-drone-completed-first-aerial-package-delivery-test-safe-drive-doorst ep-capability/ (accessed on 30 December 2022).
- Moscaritolo, A. Airblock Drone Can Turn into a Hovercraft. Available online: https://www.pcmag.com/news/airblock-dronecan-turn-into-a-hovercraft (accessed on 30 December 2022).
- Zhao, M.; Anzai, T.; Shi, F.; Chen, X.; Okada, K.; Inaba, M. Design, modeling, and control of an aerial robot dragon: A dual-rotorembedded multilink robot with the ability of multi-degree-of-freedom aerial transformation. *IEEE Robot. Autom. Lett.* 2018, 3, 1176–1183. [CrossRef]
- Drone Can Transform into a Tiny Car to Slide under Small Gaps | New Scientist. Available online: https://www.newscientist.c om/article/2204184-drone-can-transform-into-a-tiny-car-to-slide-under-small-gaps/ (accessed on 30 December 2022).

- 80. Falanga, D.; Kleber, K.; Mintchev, S.; Floreano, D.; Scaramuzza, D. The foldable drone: A morphing quadrotor that can squeeze and fly. *IEEE Robot. Autom. Lett.* **2018**, *4*, 209–216. [CrossRef]
- Miki, T.; Khrapchenkov, P.; Hori, K. UAV/UGV autonomous cooperation: UAV assists UGV to climb a cliff by attaching a tether. In Proceedings of the 2019 International Conference on Robotics and Automation (ICRA), Montreal, QC, Canada, 20–24 May 2019; pp. 8041–8047.
- Autonomous Drone-Launching Security UGV Under Development | Unmanned Systems Technology. Available online: https: //www.unmannedsystemstechnology.com/2020/11/autonomous-drone-launching-security-ugv-under-development/ (accessed on 30 December 2022).
- Russia's Autonomous Robot Tank Passes New Milestone (And Launches Drone Swarm). Available online: https://www.forbes .com/sites/davidhambling/2021/09/02/russias-autonomous-robot-tank-passes-new-milestone-and-launches-drone-swarm/?s h=2d9ae0ba21fa (accessed on 30 December 2022).
- 84. Spot Robot Dog Expands with Arm Attachment, Teams Up With Drone, and More!—Construction Junkie. Available online: https://www.constructionjunkie.com/blog/2020/12/15/spot-robot-dog-expands-with-arm-attachment-teams-up-with-droneand-more (accessed on 30 December 2022).
- 85. Researchers Create Drone that Can Land on Moving Vehicle—Autoevolution. Available online: https://www.autoevolution.com/news/researchers-create-drone-that-can-land-on-moving-vehicle-video-85516.html (accessed on 30 December 2022).
- Cocchioni, F.; Pierfelice, V.; Benini, A.; Mancini, A.; Frontoni, E.; Zingaretti, P.; Ippoliti, G.; Longhi, S. Unmanned ground and aerial vehicles in extended range indoor and outdoor missions. In Proceedings of the 2014 International conference on unmanned aircraft systems (ICUAS), Orlando, FL, USA, 27–30 May 2014; pp. 374–382.
- Saska, M.; Krajnik, T.; Pfeucil, L. Cooperative μUAV-UGV autonomous indoor surveillance. In Proceedings of the International Multi-Conference on Systems, Signals & Devices, Chemnitz, Germany, 20–23 March 2012; pp. 1–6.
- Palafox, P.R.; Garzón, M.; Valente, J.; Roldán, J.J.; Barrientos, A. Robust visual-aided autonomous takeoff, tracking, and landing of a small UAV on a moving landing platform for life-long operation. *Appl. Sci.* 2019, *9*, 2661. [CrossRef]
- Hood, S.; Benson, K.; Hamod, P.; Madison, D.; OKane, J.M.; Rekleitis, I. Bird's eye view: Cooperative exploration by UGV and UAV. In Proceedings of the 2017 International Conference on Unmanned Aircraft Systems (ICUAS), Miami, FL, USA, 13–16 June 2017; pp. 247–255.
- Cantieri, A.; Ferraz, M.; Szekir, G.; Antônio Teixeira, M.; Lima, J.; Schneider Oliveira, A.; Aurélio Wehrmeister, M. Cooperative UAV–UGV autonomous power pylon inspection: An investigation of cooperative outdoor vehicle positioning architecture. *Sensors* 2020, 20, 6384. [CrossRef] [PubMed]
- 91. Asadi, K.; Suresh, A.K.; Ender, A.; Gotad, S.; Maniyar, S.; Anand, S.; Noghabaei, M.; Han, K.; Lobaton, E.; Wu, T. An integrated UGV-UAV system for construction site data collection. *Autom. Constr.* **2020**, *112*, 103068. [CrossRef]
- TUAS—Tethered Unmanned Aerial System—DPI UAV Systems (dragonflypictures.com). Available online: https://www.drag onflypictures.com/products/tuav/ (accessed on 30 December 2022).
- Papachristos, C.; Tzes, A. The power-tethered UAV-UGV team: A collaborative strategy for navigation in partially-mapped environments. In Proceedings of the 22nd Mediterranean Conference on Control and Automation, Palermo, Italy, 16–19 June 2014; pp. 1153–1158.
- Nguyen, T.; Garone, E. Control of a UAV and a UGV cooperating to manipulate an object. In Proceedings of the 2016 American Control Conference (ACC), Boston, MA, USA, 6–8 July 2016; pp. 1347–1352.
- 95. Xiao, X.; Dufek, J.; Murphy, R.R. Tethered aerial visual assistance. arXiv 2020, arXiv:2001.06347.
- 96. Viegas, C.; Chehreh, B.; Andrade, J.; Lourenço, J. Tethered UAV with combined multi-rotor and water jet propulsion for forest fire fighting. *J. Intell. Robot. Syst.* 2022, 104, 1–13. [CrossRef]
- 97. Martinez-Rozas, S.; Alejo, D.; Caballero, F.; Merino, L. Path and trajectory planning of a tethered UAV-UGV marsupial robotics system. *arXiv* **2022**, arXiv:2204.01828.
- Papachristos, C.; Zikou, L.; Tzes, A. Deployment of a VSTOL-UAV for perception purposes in search and rescue operations. In Proceedings of the SafeChania 2015: The Knowledge Triangle in the Civil Protection Service (Education, Research, Innovation), Chania, Greece, 10–15 June 2015.
- Mullens, K.; Burmeister, A.; Wills, M.; Nelson, T.; Denewiler, T. Development of a ugv-mounted automated refueling system for VTOL UAVS. In Proceedings of the Unmanned Systems Technology VIII, Kissimmee, FL, USA, 17–20 April 2006.
- Borgese, A.; Guastella, D.C.; Sutera, G.; Muscato, G. Tether-Based Localization for Cooperative Ground and Aerial Vehicles. IEEE Robot. Autom. Lett. 2022, 7, 8162–8169. [CrossRef]
- Quaglia, G.; Visconte, C.; Scimmi, L.S.; Melchiorre, M.; Cavallone, P.; Pastorelli, S. Design of a UGV powered by solar energy for precision agriculture. *Robotics* 2020, 9, 13. [CrossRef]
- Gu, B.W.; Choi, S.Y.; Choi, Y.S.; Cai, G.; Seneviratne, L.; Rim, C.T. Novel roaming and stationary tethered aerial robots for continuous mobile missions in nuclear power plants. *Nucl. Eng. Technol.* 2016, 48, 982–996. [CrossRef]
- Drone Docking Stations | Drone Ports, Landing Pads & Hangers for UAS. Available online: https://www.unmannedsystemstec hnology.com/expo/drone-docking-stations/ (accessed on 30 December 2022).
- 104. Autonomous Drone Station (encata.net). Available online: https://www.encata.net/projects/autonomous-drone-station (accessed on 30 December 2022).

- 105. The Dronebox is a Solar-Powered Charging Station for Drones (digitalspy.com). Available online: https://www.digitalspy.com /tech/a785696/the-dronebox-is-a-solar-powered-charging-station-for-drones/ (accessed on 30 December 2022).
- Company Says Its Drone Delivery Stations Will Be Essential For Last-Mile Delivery—Inside Unmanned Systems. Available online: https://insideunmannedsystems.com/company-says-its-drone-delivery-stations-will-be-essential-for-last-mile-deliver y/ (accessed on 30 December 2022).
- Strix Drones Drone Agnostic Docking Stations DRONELIFE. Available online: https://dronelife.com/2022/05/06/strix-dronesdrone-agnostic-docking-stations-for-advanced-operations/ (accessed on 30 December 2022).
- 108. Matternet Drone Delivery: An End to End Solution—Flykit Blog. Available online: https://blog.flykit.app/matternet-drone-deli very-end-to-end/ (accessed on 30 December 2022).
- 109. Security Drone MOBOTIX AG. Available online: https://www.mobotix.com/en/partner-society/new-r-drone (accessed on 30 December 2022).
- DJI Dock: An in-House DJI Drone-in-a-Box Product at Last (Dronedj.com). Available online: https://dronedj.com/2022/03/21 /dji-dock-an-in-house-dji-drone-in-a-box-product-at-last/ (accessed on 30 December 2022).
- 111. Israeli Drones to Help Florida's Emergency Response to Hurricane Season | The Times of Israel. Available online: https://www. timesofisrael.com/israeli-drones-to-help-floridas-emergency-response-to-hurricane-season/ (accessed on 30 December 2022).
- 112. SkyX Expands with New Aircraft, Drone Charging Stations|General Aviation News: Aviation International News (ainonline.com). Available online: https://www.ainonline.com/aviation-news/general-aviation/2018--07-17/skyx-expands-new-aircra ft-drone-charging-stations (accessed on 30 December 2022).
- 10 Best DJI-Compatible Drone Docking Stations To Consider for Autonomy (flytnow.com). Available online: https://www.flyt now.com/blog/dji-compatible-docking-stations (accessed on 30 December 2022).
- 114. Switzerland's New Medical Drones | Time. Available online: https://time.com/4960622/switzerlands-new-medical-drones/ (accessed on 30 December 2022).
- Delivery Drone Station Rigged 3D—TurboSquid 1603549. Available online: https://www.turbosquid.com/3d-models/deliverydrone-station-rigged-3d-1603549 (accessed on 30 December 2022).
- Automated Drone Ground Stations by Airmada—UAS VISION. Available online: https://www.uasvision.com/2017/01/17/automated-drone-ground-stations-by-airmada/ (accessed on 30 December 2022).
- 117. HOME-Strix Drones. Available online: https://www.strixdrones.com/ (accessed on 30 December 2022).
- Heinz Schuller Vtrus ABI Zero Drone & Base Station (artstation.com). Available online: https://causticphoton.artstation.com /projects/Ya843X (accessed on 30 December 2022).
- 119. Autel now Offers an Automatic Charging Station for EVO II Drones (dronedj.com). Available online: https://dronedj.com/2021 /09/29/autel-evo-nest-drone-charging-station/ (accessed on 30 December 2022).
- 120. Avy Dock | Drone Station | Avy. Available online: https://avy.eu/our-integrated-solution/avy-dock/ (accessed on 30 December 2022).
- 121. DHL Express Launches Its First Regular Fully–Automated and Intelligent Urban Drone Delivery Service DHL Global. Available online: https://www.dhl.com/global-en/home/press/press-archive/2019/dhl-express-launches-its-first-regular-fully-automated-and-intelligent-urban-drone-delivery-service.html (accessed on 30 December 2022).
- 122. Skeyetech–Autonomous Drone for Security and Safety–Azur Drones. Available online: https://www.azurdrones.com/produ ct/skeyetech/ (accessed on 30 December 2022).
- 123. Airobotics Battery-Swapping Platform Keeps Drones Flying Around the Clock (newatlas.com). Available online: https://newatlas.com/airobotics-system-drones/43985/ (accessed on 30 December 2022).
- 124. Hey Drone Industry: Quit Griping, it's Time to Work With The FAA | ZDNET. Available online: https://www.zdnet.com/article/ hey-drone-industry-time-to-work-with-the-faa/ (accessed on 30 December 2022).
- How Drone Technologies Are Used in the Current COVID-19 Medical Response Operation PX4 Autopilot. Available online: ht tps://px4.io/how-drone-technologies-are-used-in-the-current-covid-19-medical-response-operation/ (accessed on 30 December 2022).
- 126. Skycharge Launches Drone Charging Port, SKYPORT DP5—Inside Unmanned Systems. Available online: https://insideunma nnedsystems.com/skycharge-launches-drone-charging-port-skyport-dp5/ (accessed on 30 December 2022).
- 127. Foxit Response FOXIT. Available online: https://foxit.co.za/foxit-mini/ (accessed on 30 December 2022).
- 128. FlytLaunch: Airscort FlytMini (flytnow.com). Available online: https://www.flytnow.com/flytlaunch/airscort-flytmini (accessed on 30 December 2022).
- 129. 드론용 자동충전 스테이션 I(주)아르고스다인 (argosdyne.com). Available online: https://www.argosdyne.com/en/stations (accessed on 30 December 2022).
- John Deere Shows Autonomous Sprayers Profi. Available online: https://www.profi.co.uk/news/john-deere-shows-autonomo us-sprayers/ (accessed on 30 December 2022).
- 131. Singapore Bets Big on Drones—UAS VISION. Available online: https://www.uasvision.com/2018/12/03/singapore-bets-big-on -drones/ (accessed on 30 December 2022).
- 132. Fujii, K.; Higuchi, K.; Rekimoto, J. December. Endless flyer: A continuous flying drone with automatic battery replacement. In Proceedings of the 2013 IEEE 10th International Conference on Ubiquitous Intelligence and Computing and 2013 IEEE 10th International Conference on Autonomic and Trusted Computing, IEEE, Sorrento Peninsula, Italy, 18–21 December 2013; pp. 216–223.

- 133. Nutting, J.; Mercik, K. Automated Battery Charger. Bachelor's Thesis, Worcester Polytechnic Institute, Worcester, MD, USA, April 2015.
- 134. Barrett, É.; Reiling, M.; Mirhassani, S.; Meijering, R.; Jager, J.; Mimmo, N.; Callegati, F.; Marconi, L.; Carloni, R.; Stramigioli, S. Autonomous Battery Exchange of UAVs with a Mobile Ground Base. In Proceedings of the 2018 IEEE International Conference on Robotics and Automation (ICRA), Brisbane, Australia, 21–25 May 2018; pp. 699–705.
- 135. Suzuki, K.A.; Kemper Filho, P.; Morrison, J.R. Automatic battery replacement system for UAVs: Analysis and design. *J. Intell. Robot. Syst.* **2012**, *65*, 563–586. [CrossRef]
- 136. Herath, H.M.C.W.B.; Herath, H.M.S.; Sumangala, S.W.; de Silva, O.; Chathuranga, D.; Lalitharatne, T.D. Design and development of an automated battery swapping and charging station for Multirotor Aerial Vehicles. In Proceedings of the 2017 17th International Conference on Control, Automation and Systems (ICCAS), Ramada Plaza, Jeju, Republic of Korea, 18–21 October 2017; pp. 356–361.
- Mirzaeinia, A.; Hassanalian, M.; Lee, K. Drones for Borders Surveillance: Autonomous Battery Maintenance Station and Replacement for Multirotor Drones. In Proceedings of the AIAA SciTech 2020, Orlando, FL, USA, 4–5 January 2020.
- Mourgelas, C.; Kokkinos, S.; Milidonis, A.; Voyiatzis, I. Autonomous drone charging stations: A survey. In Proceedings of the 24th Pan-Hellenic Conference on Informatics, Athens, Greece, 20–22 November 2020; pp. 233–236.
- Bin Junaid, A.; Konoiko, A.; Zweiri, Y.; Sahinkaya, M.N.; Seneviratne, L. Autonomous wireless self-charging for multi-rotor unmanned aerial vehicles. *Energies* 2017, 10, 803. [CrossRef]
- Skycharge Autonomouns Charging for Drones and Robots. Available online: https://www.skycharge.de/ (accessed on 30 December 2022).
- 141. WiBotic Autonomous Charging. Available online: https://www.wibotic.com/ (accessed on 30 December 2022).
- 142. Galeone, A. Design and Test of a Multi-Agent Robotic System for Inspection: Coordination and Integration of a Drone and a Robotic Rover. Master's Thesis, School of Mechanical and Manufacturing Engineering (SMME), Politecnico di Torino, Torino, Italy, 2019.
- Hament, B.; Oh, P. Unmanned aerial and ground vehicle (UAV-UGV) system prototype for civil infrastructure missions. In Proceedings of the 2018 IEEE International Conference on Consumer Electronics (ICCE), Berlin, Germany, 12–14 January 2018; pp. 1–4.
- 144. First Ever Flying Robot UGV Launched | Joint Forces News (joint-forces.com). Available online: https://www.joint-forces.com/d efence-equipment-news/52110-first-ever-flying-robot-ugv-launched (accessed on 30 December 2022).
- 145. Robot Dog Armed with Machine Gun Airdropped from Drone in Viral Video (taskandpurpose.com). Available online: ht tps://taskandpurpose.com/tech-tactics/robot-dog-machine-gun-drone-airdrop-kestrel-defense-video/ (accessed on 30 December 2022).
- Video Friday: Robogami, Flying Snake Robots, and Autonomous Car Eclipse—IEEE Spectrum. Available online: https://spectrum.ieee.org/video-friday-082517 (accessed on 30 December 2022).
- 147. Kawasaki's Supercharged Cargo Copter Tests Robotic Ground Crew (newatlas.com). Available online: https://newatlas.com/d rones/kawasaki-supercharged-cargo-drone-robots/ (accessed on 30 December 2022).
- 148. Nagatani, K.; Akiyama, K.; Yamauchi, G.; Otsuka, H.; Nakamura, T.; Kiribayashi, S.; Yoshida, K.; Hada, Y.; Fujino, K.; Izu, T.; et al. Volcanic ash observation in active volcano areas using teleoperated mobile robots-Introduction to our robotic-volcanoobservation project and field experiments. In Proceedings of the 2013 IEEE International Symposium on Safety, Security, and Rescue Robotics (SSRR), Linköping, Sweden,, 21–26 October 2013; pp. 1–6.
- 149. Trotter, D.A. The Lighting of Underground Mines; CRC Press: Boca Raton, FL, USA, 1983.
- 150. Lewis, W.H. Underground Coal Mine Lighting Handbook (in Two Parts): 2. Application; U.S. Department of the Interior, Bureau of Mines: Washington, DC, USA, 1986.
- Papachristos, C.; Khattak, S.; Mascarich, F.; Alexis, K. Autonomous navigation and mapping in underground mines using aerial robots. In Proceedings of the 2019 IEEE Aerospace Conference, Big Sky, MT, USA, 2–9 March 2019; pp. 1–8.
- 152. Mascarich, F.; Khattak, S.; Papachristos, C.; Alexis, K. A multi-modal mapping unit for autonomous exploration and mapping of underground tunnels. In Proceedings of the 2018 IEEE Aerospace Conference, Big Sky, MT, USA, 3–10 March 2018; pp. 1–7.
- MSHA. Historical Summary of Mine Disasters in the United States. Volume I: Coal Mines; US Department of Labor, Mine Safety and Health Administration, National Mine Health and Safety Academy: Beckley, WV, USA, 1998; pp. 1810–1958.
- 154. Available online: http://www.crushingnquarrying.com/2019/03/15/lighting-in-mines-the-not-so-dark-side (accessed on 27 March 2022).
- 155. Reddy, A.H.; Kalyan, B.; Murthy, C.S. Mine rescue robot system-a review. Procedia Earth Planet. Sci. 2015, 11, 457–462. [CrossRef]
- 156. Ohradzansky, M.T.; Rush, E.R.; Riley, D.G.; Mills, A.B.; Ahmad, S.; McGuire, S.; Biggie, H.; Harlow, K.; Miles, M.J.; Frew, E.W.; et al. Multi-agent autonomy: Advancements and challenges in subterranean exploration. *arXiv* 2021, arXiv:2110.04390. [CrossRef]
- 157. Kasper, M.; McGuire, S.; Heckman, C. A benchmark for visual-inertial odometry systems employing onboard illumination. In Proceedings of the 2019 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), Macau, China, 4–8 November 2019; pp. 5256–5263.

- Khattak, S.; Nguyen, H.; Mascarich, F.; Dang, T.; Alexis, K. Complementary multi-modal sensor fusion for resilient robot pose estimation in subterranean environments. In Proceedings of the 2020 International Conference on Unmanned Aircraft Systems (ICUAS), Athens, Greece, 1–4 September 2020; pp. 1024–1029.
- 159. Kyriazi, N.; Shubilla, J.P. Self-Contained Self-Rescuer Field Evaluation: Fourth-Phase Results; USA Department of the Interior, Bureau of Mines: Pittsburgh, PA, USA, 1994.
- 160. Aziz, N.I.; Baafi, E.Y.; MacKenzie-Wood, P. Deployment of self-contained self rescuers in Australian coal mines. In Proceedings of the 8th US Mine Ventilation Symposium, Rolla, MO, USA, 1 July 1999.
- 161. Brake, D.J.; Bates, G.P. Criteria for the design of emergency refuge stations for an underground metal mine. *Proc. AusIMM* **1999**, 304, 1–8.
- 162. Rhinehart, L. Workers at Risk: The Unfulfilled Promise of the Occupational Safety and Health Act. *West Va. Law Review.* **2008**, *111*, 117.
- Martell, M.J.; Sammarco, J.J.; Macdonald, B.; Rubinstein, E. Detectability of a self-illuminating lifeline for self-escape in smoke conditions of an underground mine. *Light. Res. Technol.* 2020, 52, 64–78. [CrossRef]
- 164. Kissell, F.N.; Timko, R.J.; Litton, C.D. Ranking factors impacting survival during coal mine fires. *Min. Eng.* 1993, 45, 1077–1083.
- Brenkley, D.; Bennett, S.C.; Jones, B. Enhancing mine emergency response. In Proceedings of the 28th International Conference on Safety in Mines Research Institutes, Sinaia, Romania, 7–11 June 1999.
- Gouws, M.J.; Phillips, H.R. Post-explosion guidance systems. In Proceedings of the 26th International Conference of Safety in Mines Research Institutes, Morgantown, WV, USA, 7 September 1995; pp. 4–8.
- 167. Harteis, S.P.; Alexander, D.W.; Chasko, L.L.; Slaughter, C.J. Evaluation of devices to enhance miner self-escape in smoke-filled entries. In Proceedings of the Society for Mining, Metallurgy, and Exploration (SME) Annual Meeting, Denver, CO, USA, 28 February–2 March 2011; p. 11–001.
- Conti, R.S. Responders to underground mine fires. In Proceedings of the 32nd Annual Conference of the Institute on Mining Health, Safety and Research, Salt Lake City, UT, USA, 5–7 August 2001.
- Kennedy, W.D.; Bayne, M.L.; Sanchez, J.G. The Use of Inflatable Barriers and Jet Fans for the Control of Smoke in the Washington Subway. In *Aerodynamics and Ventilation of Vehicles Tunnels*; Elsevier Science Publisher Limited: London, UK, 1991; pp. 433–464.
- 170. Dobroski, H., Jr.; Stolarczyk, L.G. Medium frequency radio communication system for mine rescue. In *Postdisaster Survival and Rescue Research*; US Department of the Interior, USBM Technology Transfer Seminar: Pittsburgh, PA, USA, 1982.
- 171. Raghuram, P.; Venkatesh, V. Enhancing mine safety with wireless sensor networks using zigbee technology. J. Theor. Appl. Inf. Technol. 2012, 37, 261–267.
- 172. Sommer, C.; Dressler, F. Vehicular Networking; Cambridge University Press: Cambridge, UK, 2015.
- Araniti, G.; Campolo, C.; Condoluci, M.; Iera, A.; Molinaro, A. LTE for vehicular networking: A survey. *IEEE Commun. Mag.* 2013, 51, 148–157. [CrossRef]
- Kim, Y.H.; Cahyadi, W.A.; Chung, Y.H. Experimental demonstration of VLC-based vehicle-to-vehicle communications under fog conditions. *IEEE Photonics J.* 2015, 7, 1–9. [CrossRef]
- 175. Abualigah, L.; Diabat, A.; Sumari, P.; Gandomi, A.H. Applications, deployments, and integration of internet of drones (IoD): A review. *IEEE Sens. J.* 2021, 21, 25532–25546. [CrossRef]
- 176. Bekmezci, I.; Sahingoz, O.K.; Temel, Ş. Flying ad-hoc networks (FANETs): A survey. Ad. Hoc. Netw. 2013, 11, 1254–1270. [CrossRef]
- 177. Wang, J.; Jiang, C. Flying Ad Hoc Networks: Cooperative Networking and Resource Allocation; Springer Nature: Berlin/Heidelberg, Germany, 2022.
- 178. Campion, M.; Ranganathan, P.; Faruque, S. UAV swarm communication and control architectures: A review. *J. Unmanned Veh. Syst.* **2018**, *7*, 93–106. [CrossRef]
- 179. Roberge, V.; Tarbouchi, M.; Labonté, G. Comparison of parallel genetic algorithm and particle swarm optimization for real-time UAV path planning. *IEEE Trans. Ind. Inform.* **2012**, *9*, 132–141. [CrossRef]
- Duan, H.; Luo, Q.; Shi, Y.; Ma, G. Hybrid particle swarm optimization and genetic algorithm for multi-uav formation reconfiguration. *IEEE Comput. Intell. Mag.* 2013, *8*, 16–27. [CrossRef]
- 181. Rosati, S.; Krużelecki, K.; Heitz, G.; Floreano, D.; Rimoldi, B. Dynamic routing for flying ad hoc networks. *IEEE Trans. Veh. Technol.* 2015, *65*, 1690–1700. [CrossRef]
- 182. Bekmezci, İ.; Şentürk, E.; Türker, T. Security issues in flying ad-hoc networks (FANETS). J. Aeronaut. Space Technol. 2016, 9, 13–21.
- Zafar, W.; Khan, B.M. Flying ad-hoc networks: Technological and social implications. *IEEE Technol. Soc. Mag.* 2016, 35, 67–74. [CrossRef]
- Sun, J.; Wang, W.; Kou, L.; Lin, Y.; Zhang, L.; Da, Q.; Chen, L. A Data Authentication Scheme for UAV ad hoc network communication. J. Supercomput. 2020, 76, 4041–4056. [CrossRef]
- 185. Lv, Z. The security of Internet of Drones. Comput. Commun. 2019, 148, 208–214. [CrossRef]
- Wu, T.; Guo, X.; Chen, Y.; Kumari, S.; Chen, C. Amassing the security: An enhanced authentication protocol for drone communications over 5G networks. *Drones* 2022, 6, 10. [CrossRef]
- Bunse, C.; Plotz, S. Security analysis of drone communication protocols. In Proceedings of the Engineering Secure Software and Systems: 10th International Symposium, ESSoS 2018, Paris, France, 26–27 June 2018.
- 188. Long, T.; Ozger, M.; Cetinkaya, O.; Akan, O.B. Energy neutral Internet of Drones. IEEE Commun. Mag. 2018, 56, 22–28. [CrossRef]

- 189. Bor-Yaliniz, I.; Salem, M.; Senerath, G.; Yanikomeroglu, H. Is 5G Ready for Drones: A look into contemporary and prospective wireless networks from a standardization perspective. *IEEE Wirel. Commun.* **2019**, *26*, 18–27. [CrossRef]
- 190. Available online: https://arlweb.msha.gov/TECHSUPP/acc/application/acri2011.pdf (accessed on 8 February 2023.).
- 191. Mohsan, S.A.H.; Khan, M.A.; Noor, F.; Ullah, I.; Alsharif, M.H. Towards the unmanned aerial vehicles (UAVs): A comprehensive review. *Drones* 2022, *6*, 147. [CrossRef]
- Dubaniewicz, T.H.; DuCarme, J.P. Are lithium ion cells intrinsically safe? *IEEE Trans. Ind. Appl.* 2013, 49, 2451–2460. [CrossRef]
 [PubMed]
- Bergveld, H.J.; Kruijt, W.S.; Notten, P.H. Battery management systems. In *Battery Management Systems*; Springer: Dordrecht, The Netherlands, 2002; pp. 9–30.
- 194. USA Department of Labor. PROGRAM INFORMATION BULLETIN; NO. P11-1; 23 March 2011.
- Fotouhi, A.; Auger, D.J.; O'Neill, L.; Cleaver, T.; Walus, S. Lithium-sulfur battery technology readiness and applications—A review. *Energies* 2017, 10, 1937. [CrossRef]
- 196. Cuffari, B. Improving the Safety of Batteries in Underground Coal Mines. Editorial Feature, p. 2. 17 October 2017. Available online: https://www.azomining.com/Article.aspx?ArticleID=1366 (accessed on 5 December 2022).
- 197. Dvorznak, G.J. A Permissibility Checklist for Approved Diesel Powered Coal Mining Equipment. U.S. Department of Labor, Mine Safety and Health Administration, Triadelphia, West Virginia. Available online: https://citeseerx.ist.psu.edu/document?r epid=rep1&type=pdf&doi=0bcc9277062e24c29eb702907816ef203be01832 (accessed on 5 December 2022).
- 198. Baldic, J.; Osenar, P.; Lauder, N.; Launie, P. Fuel cell systems for long duration electric UAVs and UGVs. In Proceedings of the Defense Transformation and Net-Centric Systems 2010, Orlando, FL, USA, 6–8 April 2010; Volume 7707, pp. 9–17.
- Lee, J.; Wettergreen, D.; Kantor, G. Lightweight Laser Scan Registration in Underground Mines with Band-based Downsampling Method. In *Field and Service Robotics*; Yoshida, K., Tadokoro, S., Eds.; Springer Tracts in Advanced Robotics, vol 92; Springer: Berlin/Heidelberg, Germany, 2014. [CrossRef]
- Nagatani, K.; Otake, K.; Yoshida, K. Three-Dimensional Thermography Mapping for Mobile Rescue Robots. In *Field and Service Robotics*; Yoshida, K., Tadokoro, S., Eds.; Springer Tracts in Advanced Robotics, vol 92; Springer: Berlin/Heidelberg, Germany, 2014. [CrossRef]
- Lavigne, N.J.; Marshall, J.A. A landmark–bounded method for large–scale underground mine mapping. J. Field Robot. 2012, 29, 861–879. [CrossRef]
- Neumann, D.I.K.; Berg, D.I.J.; Möllemann, G.; Nienhaus, D.I.K. Collision Avoidance System for the Underground Mining Industry. *IFAC-Pap.* 2015, 48, 60–65. [CrossRef]
- 203. Saeed, A.S.; Younes, A.B.; Cai, C.; Cai, G. A survey of hybrid unmanned aerial vehicles. *Prog. Aerosp. Sci.* 2018, 98, 91–105. [CrossRef]
- 204. Yao, H.; Qin, R.; Chen, X. Unmanned aerial vehicle for remote sensing applications—A review. *Remote Sens.* 2019, *11*, 1443. [CrossRef]
- Jones, E.; Sofonia, J.; Canales, C.; Hrabar, S.; Kendoul, F. Applications for the Hovermap autonomous drone system in underground mining operations. J. S. Afr. Inst. Min. Metall. 2020, 120, 49–56. [CrossRef]
- 206. Tranzatto, M.; Miki, T.; Dharmadhikari, M.; Bernreiter, L.; Kulkarni, M.; Mascarich, F.; Andersson, O.; Khattak, S.; Hutter, M.; Siegwart, R. CERBERUS in the DARPA Subterranean Challenge. *Sci. Robot.* 2022, *7*, eabp9742. [CrossRef] [PubMed]
- 207. Hudson, N.; Talbot, F.; Cox, M.; Williams, J.; Hines, T.; Pitt, A.; Wood, B.; Frousheger, D.; Surdo, K.L.; Molnar, T. Heterogeneous ground and air platforms, homogeneous sensing: Team CSIRO Data61's approach to the DARPA subterranean challenge. *arXiv* 2021, arXiv:2104.09053. [CrossRef]
- 208. Roucek, T.; Pecka, M.; Cızek, P.; Petricek, T.; Bayer, J.; Šalansky, V.; Azayev, T.; Hert, D.; Petrlik, M.; Báca, T. System for multi-robotic exploration of underground environments CTU-CRAS-NORLAB in the DARPA subterranean challenge. arXiv 2021, arXiv:2110.05911.
- 209. Agha, A.; Otsu, K.; Morrell, B.; Fan, D.D.; Thakker, R.; Santamaria-Navarro, A.; Kim, S.-K.; Bouman, A.; Lei, X.; Edlund, J. Nebula: Quest for robotic autonomy in challenging environments; team CoSTAR at the DARPA subterranean challenge. *arXiv* 2021, arXiv:2103.11470.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.